

MIT Center for Energy and Environmental Policy Research

Working Paper Series

Optimality Conditions and Cost Recovery in Electricity Markets with Variable Renewable Energy and Energy Storage

MAGNUS KORPÅS AND AUDUN BOTTERUD

MARCH 2020

CEEPR WP 2020-005



MASSACHUSETTS INSTITUTE OF TECHNOLOGY

Optimality Conditions and Cost Recovery in Electricity Markets with Variable Renewable Energy and Energy Storage

Magnus Korpås¹, Audun Botterud²

¹Dept. of Electric Power Engineering, Norwegian University of Science and Technology

²Laboratory of Information and Decision Systems, MIT, and Energy Systems Division, Argonne National Laboratory.

Abstract

We formulate generation capacity portfolio planning in the power grid as a least-cost optimization problem and derive analytical expressions for the optimality conditions for dispatchable generation, variable renewable energy (VRE), and energy storage systems (EES) using a generalized net load duration curve approach. This is done for different operational strategies for EES with and without VRE in the system. For all studied combinations of technologies and operational strategies, we show that all units, including VRE and EES, recover their costs and maximize their profits in the system optimum, for an ideal short-term electricity market based on marginal cost and scarcity pricing. We verify the analytical findings through a numerical example, which shows that the general net load duration curve approach gives identical results to a standard capacity expansion model with sequential operation of the generation and ESS units, under the assumption of limited power capacity but infinite energy capacity of EES. The results highlight that the net load duration curve models presented in this paper can be a useful supplement to more detailed simulation studies of markets with high penetration of VRE and EES, to better understand the underlying factors that determines the optimal capacity mix and profitability of each technology in energy-only electricity markets.

Keywords: electricity markets, optimality conditions, market equilibrium, variable renewable energy, energy storage system, duration curve model

Corresponding Author: Magnus Korpås Tel: +47 97 04 20 09 E-mail Address: magnus.korpas@ntnu.no Postal Address: Dept. of Electric Power Engineering, NTNU, NO-7491 Trondheim, Norway

Nomenclature

Indices	
b	Baseload plant
d	Demand
е	Electric Energy Storage (EES)
e+	Discharging of EES
e-	Charging of EES
F	Firm
G	(Thermal) generator
nd	Net demand
Р	Peaker plant
S	Load shedding
V	Variable renewable energy (VRE)
vpot	VRE potential (uncurtailed)
Symbols	
α_i	Annuity factor of plant <i>i</i> [p.u.]
η_i	Efficiency of plant <i>i</i> [p.u.]
λ_d	Lagrange multiplier for power balance [\$/MWh]
μ_i^{min}	Lagrange multiplier for minimum generation of plant <i>i</i>
μ_i^{max}	Lagrange multiplier for maximum generation of plant <i>i</i>
π_i	Profit function for plant <i>i</i> [\$/yr]
Ĺ	Lagrangian
d	Storage duration [h]
p	Price of electricity in the market [\$/MWh]
p_{CO2}	Price for CO2 emissions [E/tonco2]
$\frac{q}{a}$	Maximum/minimum generation or consumption [MW]
<i>q</i> , <u>q</u>	
R	Discount rate [%]
t	Duration of time step [n] Variable cost of plant i [\$/MWh]
v_i	Variable cost of plant i [3/10/10/11] Power capacity of plant i [MW]
ACE	A verge cost of electricity $[\%/WWh]$
AC	Annual cost of plant i [\$/vr]
AR_i	Annual revenue of plant $i [\$/yr]$
AE.	Availability factor of VRE plant [p_{μ}]
$A F^{[t_{i-1},t_i]}$	Availability factor of VRE plant during the time segment between t_{i-1} and t_i [p.u.]
C	Total annual system costs $[\$/vr]$
CRF	Cost recovery factor [1/yr]
Fi	Annual fixed costs of plant <i>i</i> [\$/MW/vr]
F_{-}^{pwr}	Annual fixed cost of EES power capacity [\$/kW/vr]
F_{2}^{en}	Annual fixed cost of EES energy capacity [\$/kWh/yr]
Ĕ	Energy [MWh]
EFfuel	Emission factor of fuel [ton _{CO2} /MWh _{fuel}]
\overline{E}_{-}	Energy capacity of EES [MWh]
L	Lifetime [vr]
LCOE	Levelized cost of electricity [\$/MWh]
OM _{mar} i	Variable operation and maintenance cost of plant <i>i</i> [\$/MWh]
OM fires	Fixed operation and maintenance cost of plant <i>i</i> [\$/kW/vr]
SCC_i	Specific capital cost of plant i [\$/kW/vr]
T	Hours of the year (T=8760 h)
WAPE	Weighted average price of electricity [\$/MWh]
Sets	
Seis	

Ε	EES plants
G	(Thermal) generators
V	VRE plants

1 Introduction

Variable Renewable Energy (VRE) technologies are now deployed at an accelerated phase in electricity markets all over the world. Up to now, investments in these resources have to a large extent been driven by different subsidy schemes, either in the form of feed-in tariffs, green certificates, compulsory power purchasing agreements (PPAs), investment refunds, tax credits or other mechanisms. However, in the last few years, the cost for field-ready VRE installations has declined so fast that in more and more areas, onshore wind and utility-scale solar have reached a cost-level that is lower than conventional generation without any form of subsidy (Smith 2019).

In a future where VRE is the cheapest technology, calculated in terms of lifetime costs of kWh delivered, current electricity markets are challenged if no changes to their design takes place. First and foremost, this is due to the well-known merit-order effect (Sensfuß, Ragwitz, and Genoese 2008; Trötscher and Korpås 2008; Hirth 2013; Levin and Botterud 2015), which simply expresses that conventional generators with higher marginal costs are dispatched less and potentially pushed out of the market as more low or zero marginal cost VRE enters the system and reduce the average short-term price in the electricity market.

In principle, the merit-order effect will also impact the deployment of VRE as long as they receive no subsidies and must rely on electricity short-term prices in the electricity market to cover their expenses. Although often discussed as a challenge in the literature (Milligan et al. 2015; Pollitt and Anaya 2016; Botterud and Auer 2019), this limiting factor of VRE expansion in electricity markets has not been studied in much depth, with regards to the cost recovery conditions of the VRE plants or other generators in the system. Other relevant studies have taken a central planner approach for future scenarios (California Energy Commission 2018; European Commission 2016), or investigated present subsidy schemes for VRE (Hiroux and Saguan 2010; Kalkuhl, Edenhofer, and Lessmann 2013; Nicolini and Tavoni 2017) or capacity credits (Bothwell and Hobbs 2017). The literature has often described how externally funded VRE plants (i.e. built outside the pure market incentives) lowers the net load and thereby lowers the income of thermal generators, see (Traber and Kemfert 2011), and impact the resource adequacy in the system (Milligan et al. 2016). As recently described in (Joskow 2019), most theoretical work so far has treated VRE capacity as given exogenously. An important exception is (Helm and Mier 2016), which present a two-stage model for efficient investment in VRE and fossil generators for deceasing VRE capacity costs, assuming a uniform¹ availability function for VRE variability. In this paper, we will derive simple but generally valid cost recovery conditions for VRE and thermal generators in energy-only markets. Under a set of assumptions, we show that all generators (including VRE) recovers their costs by traditional marginal cost pricing, and that this results in an optimal generation capacity portfolio for the system. This implies that the merit-order effect of VRE may not be a problem for efficient development and operation of the power market as such, but it will obviously have an impact on the number of conventional generators that is needed in the system.

In this paper, we further investigate the market equilibrium implications of introducing energy storage systems (ESS) in energy-only markets based on marginal cost pricing. VRE, ESS, and especially batteries, have experienced a tremendous cost reduction in recent years, and there is a vast number of research articles on how EES can be used to facilitate VRE in future power systems, both from technical, economical, and environmental viewpoints, see e.g (Toledo, Oliveira Filho, and Diniz 2010; Díaz-González et al. 2012; Zhao et al. 2015; Denholm et al. 2010; Sioshansi et al. 2009). Most literature that studies the economic viability of EES either takes a system cost-minimization perspective (N. Li et al. 2016; Arbabzadeh et al. 2019; de Sisternes, Jenkins, and Botterud 2016) or a price-taker perspective (Bathurst and Strbac 2003; Korpås, Holen, and Hildrum 2003; Sioshansi et al. 2009). To our knowledge, there is not much published material on the equilibrium conditions for EES in ideal energy-only markets based on marginal cost pricing. Green and Staffell (2015) analyze the impact of ESS on wholesale markets with large amount of renewable energy using a cost-minimizing optimization model. They show that ESS reduces the price volatility and some of the very highest prices, but also leads to higher near-peak prices due to the influence on the equilibrium condition of the conventional generators. They

¹ Their model is not limited to uniform functions, but this representation is chosen for convenience in order to derive closed-form solutions.

also describe that ESS increases the market value of wind power by raising the prices in the low-price periods and reduces the number of zero-price hours. Steffen and Weber (2013) establish optimality conditions for ESS in a system with thermal generators and VRE plants using a similar duration curve as here, but do not analyze market aspects or optimality conditions for the VRE power plants. A recent contribution by Schmalensee (2019) takes a theoretical approach to analyze market aspects of both ESS and VRE plants. By introducing ESS and stochastic VRE into a two-stage model, the paper suggests that the long-run equilibrium value of storage capacity minimizes expected system cost in most cases. However, the paper also states that it cannot be ruled out that inefficient equilibria exist when ESS is introduced to the system. Schmalensee models the daily operation of EES within a two-stage model, and derives first-order and second-order optimality conditions for all units. In contrast, we derive system optimality and cost recovery conditions using a model that represents hourly operation of the system over a full year based on a load duration curve approach. Helm and Mier (2018) analyze optimal subsidies and capacities for VRE and EES in systems with imperfect carbon pricing. As in (Schmalensee 2019), they derive optimality conditions over a representative storage cycle for an assumed regular pattern of renewable energy input.

In this paper, we take an analytical approach to study market equilibrium in competitive low-carbon electricity markets. We first derive analytical expressions for the optimality conditions for thermal generators, VRE and EES where the objective is to minimize the total cost of a system with fixed demand. We then show how profit maximization of each generation and storage resource in a market based on marginal cost pricing and administrative scarcity pricing can give the same results as the optimal investment portfolio under system cost minimization. Our approach follows to a large extent traditional literature on system optimality and cost recovery with thermal generators based on a load duration curve for demand and constant marginal generator costs (Stoft 2002; Green 2000). However, we extend the analysis to incorporate VRE and EES, and investigate the long-term market equilibrium implications that follow from different operational strategies used for the EES. Finally, the theoretical findings are illustrated by a simple, yet representative numerical example based on data and scenarios for the European power system in 2050. Although the duration curve approach is simplified with respect to representation of energy storage constraints, it is proven to be useful for illustrating the impacts of VRE and EES on market equilibrium in an easy and transparent way.

The rest of the paper is organized as follows: Chapter 2 presents the general capacity expansion problem that is used to derive the system optimality conditions in Chapters 3-4. Chapter 3 recaptures the traditional system with only thermal generators meeting the load represented as a duration curve. Chapter 4 introduces a general representation of VRE technologies in the net load duration curve framework. Chapter 5 presents the EES, which is modelled for two system configurations: Only thermal power plants (Chapter 5.1) and VRE + thermal (Chapter 5.2). In the latter case, the optimality conditions of all types of plants are derived for three different operational strategies for EES. Throughout Chapters 3-5, we verify that the optimal least-cost system solution is also obtained in a perfect market with marginal cost pricing, by investigating cost recovery conditions under profit maximization of price-taking generation units. Chapter 6 presents the results of a numeric case study based on real data and scenarios for the European power system stadium 2050. Finally, Chapter 7 concludes the paper and provides directions for future research.

2 Scope and methodology

We consider an energy-only market with scarcity pricing, i.e. an administratively determined price during supply shortages, as the basis for analysis, without discussing additional capacity remuneration mechanisms. As shown in previous literature, e.g. by (Stoft 2002) and (Green 2000), an energy-only market with scarcity pricing provides the theoretically optimal incentives for investments in dispatchable generators. It is also known that the presence of VRE changes the amount of optimally installed thermal generation in a system (Levin and Botterud 2015), but less is known about how costs are recovered in equilibrium, including costs for VRE and ESS. In our model, we treat VRE and ESS capacity as endogenous variables using a simplified yet representative model of VRE output variations, and derive analytical expressions for the equilibrium conditions for all generators given marginal cost pricing.

Demand side flexibility is expected to play a vital role in enhancing VRE integration (Strbac 2008), and the system optimization problem must therefore in the general case be formulated as a welfare maximization problem. Since the scope of this paper is cost recovery of conventional generation, VRE and ESS devices, we reformulate the system optimization problem to a deterministic cost minimization problem, and leave the inclusion of demand side flexibility and uncertainty for future work as these are substantial topics on their own. Hence, the central planner's least-cost problem can in its basic form be formulated as:

$$\min C = \sum_{g=1}^{G} \left[F_g x_g + v_g \int_{0}^{T} q_g(t) dt \right] + \sum_{\nu=1}^{V} F_{\nu} x_{\nu} + \sum_{e=1}^{E} \left[F_e^{pwr} \cdot x_e + F_e^{en} \cdot \overline{E}_e \right] + v_s \int_{0}^{T} q_s(t) dt$$
(1)

s.t.
$$\sum_{g=1}^{G} q_g(t) + \sum_{\nu=1}^{V} q_{\nu}(t) + \sum_{e=1}^{E} [q_e(t) - q_{e-}(t)] + q_s(t) = q_d(t)$$
(2)

$$0 \le q_g(t) \le x_g \ \forall \ g \in G \tag{3}$$

$$0 \le q_v(t) \le AF_v(t) \cdot x_v \quad \forall \ v \in V \tag{4}$$

$$0 \le q_{e+}(t) \le x_e \quad \forall \ e \cdot \in E \tag{5}$$

$$0 \le q_{e-}(t) \le x_e \qquad \forall \ e \cdot \in E \tag{6}$$

$$\frac{dE_e(t)}{dt} = \eta_{e-} \cdot q_{e-}(t) - \frac{q_{e+}(t)}{\eta_{e+}} \quad \forall e \in E$$
⁽⁷⁾

$$0 \le E_e(t) \le \overline{E}_e \quad \forall \ e \cdot \in E \tag{8}$$

The objective function (1) minimizes the sum of annualized fixed costs, annual variable generation costs and annual costs of load shedding. Variable costs of VRE are set to zero under the assumption that variable O&M costs are negligible. ESS investment costs are linear functions of both power capacity and energy capacity. The instantaneous power balance is given in (2), conventional generation limits in (3) while VRE output is limited by the instantaneous availability factor in (4). EES limits and storage balances, including roundtrip losses, are expressed in (5)-(8). Note that our formulation focuses on the market for energy and does not include ancillary services. Moreover, the impacts of the transmission network and generator unit commitment are also ignored.

The linear model described by (1)-(8) is applicable for any number and types of fossil generators, VRE generators and EES units. To ease the notation in the remainder of the paper, we limit the model to two types of thermal generators (i.e. peaker and baseload), one type of VRE technology and one type of EES. By "type" we here refer to a technology with a given cost and performance, i.e. two units of the same type are identical. Moreover, (8) explicitly models the limit on energy storage capacity, which is straightforward to implement in a time-sequential model, but only possible to account for indirectly in duration curve models as discussed in more detail in Chapter 5. By applying these assumptions, the cost-minimization problem can be simplified to:

$$\min_{x_i, q_k(t), q_{e^-}(t)} C = \sum_i F_i x_i + \sum_j v_j \int_0^T q_j(t) \, dt \tag{9}$$

s.t.
$$q_d(t) - \sum_k q_k(t) + q_{e-}(t) = 0$$
 (10)

$$-q_k(t) \le 0$$
, $-q_{e^-}(t) \le 0$ (11)

$$q_l(t) - x_l \le 0$$
, $q_{e^-}(t) - x_e \le 0$ (12)

$$q_{\nu}(t) - AF_{\nu}(t)x_{\nu} \le 0 \tag{13}$$

$$\eta_e \int_0^T q_{e-}(t)dt - \int_0^T q_e(t)dt = 0$$
Sets: $i \in \{p, b, v, e\}, j \in \{s, p, b\}, k \in \{s, p, b, v, e\}, l \in \{p, b, e\}$
(14)

where indices *s*, *p*, *b*, *v*, *e* refers to load shedding, peaker, baseload, VRE plant and EES respectively. The administrative price assumed to be set during load shedding has been expressed as a variable cost $v_s = p_s$ for ease of notation. Moreover, we simplify storage notation by defining $q_e \equiv q_{e+}$, so that q_e refers to discharging power, while charging is q_{e-} as before. Equation (14) is the energy storage balance over the whole analysis period, where we have introduced the round-trip efficiency $\eta_e = \eta_{e-} \cdot \eta_{e+}$.

3 Conventional generators

3.1 System optimality conditions

Optimality conditions for conventional generators in power markets have been extensively covered in the literature e.g. (Joskow 1976; Stoft 2002; Green 2000). For the sake of consistency with the later part of the paper, we show here how to derive the first-order optimality conditions for a system consisting of only conventional generators which serve a given demand with a temporal profile over the year. The system cost-minimization problem (9)-(14) becomes:

$$\min_{x_i, q_j(t)} C = \sum_i F_i x_i + \sum_j v_j \int_0^1 q_j(t) \, dt \quad \forall \, i \in \{p, b\}, j \in \{s, p, b\}$$
(15)

s.t.
$$\sum_{j} q_j(t) - q_d(t) = 0 \qquad \forall j \in \{s, p, b\} \quad (\lambda_d)$$
(16)

$$-q_j(t) \le 0 \qquad \forall j \in \{s, p, b\} \ (\mu_j^{min})$$
(17)

$$q_i(t) - x_i \le 0 \qquad \forall i \in \{p, b\} \quad (\mu_i^{max})$$
(18)

By sorting the time-varying demand over the year in descending order, we obtain the duration curve as illustrated in Figure 1. The time parameter $t \in [0, T]$ now refer to the sorted demand (i.e. the duration curve) and not the chronological time-series. The figure also displays how the demand is covered by the two generators for a solution where both generators are part of the optimal generation portfolio, i.e. x_p and x_b are both strictly positive.

Consider first the operational problem for an arbitrary time instant t and given plant capacities x_i . The investment cost term in (15) becomes 0, and we can write the Lagrangian

$$\mathcal{L}_{op}(q_{j},\lambda_{d},\mu_{j}^{min},\mu_{i}^{max},t)$$

$$= \sum_{j} v_{j}q_{j}(t) + \lambda_{d}(t) \cdot \left(q_{d}(t) - \sum_{j} q_{j}(t)\right)$$

$$+ \sum_{j} \mu_{j}^{min}(t) \cdot \left(-q_{j}(t)\right) + \sum_{i} \mu_{i}^{max}(t) \cdot \left(q_{i}(t) - x_{i}\right)$$

$$\forall i \in \{p,b\}, j \in \{s,p,b\}$$

$$(19)$$

where λ_d is the Lagrange multiplier associated with the demand balance (16), μ_j^{min} is the Lagrange multiplier for generator *j* in (17) and μ_i^{max} is the Lagrange multiplier for generator *i* in (18). Note that multiplier μ_j^{min} also applies for load shedding, i.e. for $j \in \{s, p, b\}$. λ_d is the system marginal cost, which defines the spot price for energy under marginal pricing (Hasan, Galiana, and Conejo 2008). The optimal dispatch is given by the Karush-Kuhn-Tucker (KKT) conditions which in addition to (16)-(18) comprises

$$v_i - \lambda_d(t) + \mu_i^{max}(t) - \mu_i^{min}(t) = 0 \,\forall \, i \in \{p, b\}$$
(20)

$$\nu_s - \lambda_d(t) - \mu_s^{min}(t) = 0 \tag{21}$$

$$\mu_i^{max}(t) \cdot (q_i(t) - x_i) = 0 \ \forall \ i \in \{p, b\}$$
(22)

$$\mu_j^{min}(t) \cdot \left(-q_j(t)\right) = 0 \forall j \in \{s, p, b\}$$
⁽²³⁾

where we have followed the sign convention from (Gabriel et al. 2013). It should be noted that the Lagrangian and corresponding KKT-conditions derived above apply to any number of generators in the set i, although we have used an example with only two generators here. The KKT conditions provide the optimal operation of each generator for all demand levels, as summarized for our two-generator problem in Table 1.

$t \leq t_s$:	$t_s < t \le t_p$:	$t > t_p$:
$x_b + x_p \le q_d(t)$	$x_b \le q_d(t) < x_b + x_p$	$q_d(t) > x_b$
$\lambda_d(t) = v_s$	$\lambda_d(t) = v_p$	$\lambda_d(t) = v_b$
$q_b(t) = x_b$	$q_b(t) = x_b$	$q_b(t) = q_d(t)$
$q_p(t) = x_p$	$q_p(t) = q_d(t) - x_b$	$q_p(t) = 0$
$q_s(t) = q_d(t) - x_p - x_b$	$q_s(t) = 0$	$q_s(t) = 0$

Table 1. System marginal cost and optimal operation for different demand levels.

From Table 1, we see that the output of each generator and load shedding are functions of the demand $q_d(t)$ and the installed capacities $x_i \forall i \in \{p, b\}$. For given installed capacities, we also know the output of each unit, since the units must be dispatched due to the merit order to minimize operating costs. We can therefore convert the problem (15)-(18) to an unconstrained cost minimization problem:

$$\min_{x_p, x_b} C = F_p x_p + F_b x_b + v_s \int_0^{t_s} [q_d(t) - x_p - x_b] dt + v_p \cdot \left(\int_0^{t_s} x_p dt + \int_{t_s}^{t_p} [q_d(t) - x_b] dt \right) + v_b \cdot \left(\int_0^{t_p} x_b dt + \int_{t_p}^{T} q_d(t) dt \right) = F_p x_p + F_b x_b + (24)$$

$$v_s \left(E_d^{[0,t_s]} - (x_p + x_b) t_s \right) + v_p \left(E_d^{[t_s,t_p]} + (x_p + x_b) t_s - x_b t_p \right) + v_b \left(E_d^{[t_p,T]} + x_b t_p \right)$$

where $E_d^{[t_1,t_2]}$ is the energy demand over the period t_1 to t_2 , corresponding to the area in Figure 1 bounded by $[t_1, t_2]$ and $[q_d(t_1), q_d(t_2)]$. The relation between the durations and generation capacities is also illustrated in the figure and we can write $t_p = f(x_b)$ and $t_s = f(x_p, x_b)$. Derivation of C with respect to the unknowns x_p and x_b gives:

$$\frac{\partial C}{\partial x_p} = F_p - (v_s - v_p)t_s + v_s \left(\underbrace{\frac{\partial E_d^{[0,t_s]}}{\partial x_p} - (x_p + x_b)\frac{\partial t_s}{\partial x_p}}_{+ v_p \left(\underbrace{\frac{\partial E_d^{[t_s,t_p]}}{\partial x_p} + (x_p + x_b)\frac{\partial t_s}{\partial x_p}}_{+ v_p \left(\underbrace{\frac{\partial E_d^{[t_s,t_p]}}{\partial x_p} + (x_p + x_b)\frac{\partial t_s}{\partial x_p}}_{+ v_s \left(\underbrace{\frac{\partial E_d^{[0,t_s]}}{\partial x_b} - (x_p + x_b)\frac{\partial t_s}{\partial x_b}}_{+ v_b \right)} \right)}_{+ v_p \left(\underbrace{\frac{\partial E_d^{[t_s,t_p]}}{\partial x_b} + (x_p + x_b)\frac{\partial t_s}{\partial x_b}}_{+ v_b \left(\underbrace{\frac{\partial E_d^{[t_s,t_p]}}{\partial x_b} + x_b\frac{\partial t_p}{\partial x_b}}_{+ v_b \left(\frac{\partial E_d^{[t_p,T]}}{\partial x_b} + x_b\frac{\partial t_p}{\partial x_b} \right)} \right)}_{+ v_b \left(\underbrace{\frac{\partial E_d^{[t_p,T]}}{\partial x_b} + x_b\frac{\partial t_p}{\partial x_b}}_{+ v_b \left(\frac{\partial E_d^{[t_p,T]}}{\partial x_b} + x_b\frac{\partial t_p}{\partial x_b} \right)} \right)_{+ v_b \left(\underbrace{\frac{\partial E_d^{[t_p,T]}}{\partial x_b} + x_b\frac{\partial t_p}{\partial x_b}}_{+ v_b \left(\frac{\partial E_d^{[t_p,T]}}{\partial x_b} \right)} \right)_{+ v_b \left(\frac{\partial E_d^{[t_p,T]}}{\partial x_b} + x_b\frac{\partial t_p}{\partial x_b} \right)}_{+ v_b \left(\frac{\partial E_d^{[t_p,T]}}{\partial x_b} + x_b\frac{\partial t_p}{\partial x_b} \right)} \right)_{+ v_b \left(\underbrace{\frac{\partial E_d^{[t_p,T]}}{\partial x_b} + x_b\frac{\partial t_p}{\partial x_b}}_{+ v_b \left(\frac{\partial E_d^{[t_p,T]}}{\partial x_b} + x_b\frac{\partial t_p}{\partial x_b} \right)} \right)_{+ v_b \left(\underbrace{\frac{\partial E_d^{[t_p,T]}}{\partial x_b} + x_b\frac{\partial t_p}{\partial x_b}}_{+ v_b \left(\frac{\partial E_d^{[t_p,T]}}{\partial x_b} \right)_{+ v_b \left($$

 $(v_s - v_p)t_s - (v_p - v_b)t_p$ The first-order optimality conditions become:

$$\frac{\partial C}{\partial x_p} = 0 \Rightarrow t_s = \frac{F_p}{(v_s - v_p)}$$
(27)

$$\frac{\partial C}{\partial x_b} = 0 \Rightarrow t_p = \frac{F_b - F_p}{(v_p - v_b)}$$
(28)

When the durations are known, the optimal capacities are found from the demand curve $x_b = q_d(t_p)$ and $x_p = q_d(t_s) - x_b$. Problem (24) with solutions (27) and (28) are valid under the assumption that both x_p and x_b are strictly positive. To find the true optimal solution of (15) using the duration curve method, we must compare the result of (24) with the result for all other possible generator combinations. In this case, that means to remove the peaker and the baseplant, respectively, from the solution space and obtain optimal durations for these cases.



Figure 1. Load duration curve with durations and optimal capacities of peaker and base plants.

3.2 Profit maximization and cost recovery

The annual profit of generator *i* in the market is:

$$\pi_{i} = AR_{i} - AC_{i} = \int_{0}^{t_{i}} (p(t) - v_{i})q_{i}(t)dt - F_{i}x_{i}$$
⁽²⁹⁾

The price p(t) is equal to the dual value λ_d of the load balance (16) in a system with marginal cost pricing. Using the values of λ_d and the optimal operation regimes derived in the previous section, the profit functions for the peaker and baseplant become:

$$\pi_p = (v_s - v_p)t_s x_p - F_p x_p = \left((v_s - v_p)t_s - F_p\right)x_p \tag{30}$$

$$\pi_{b} = (v_{s} - v_{b})t_{s}x_{b} + (v_{p} - v_{b})(t_{p} - t_{s})x_{b} - F_{b}x_{b}$$

$$= ((v_{s} - v_{b})t_{s} + (v_{s} - v_{b})(t_{p} - t_{s}) - F_{b})x_{b}$$
(31)

It follows that in the system optimum, the profits of the individual generators are 0, i.e. all generators recovers their cost: $\pi_p = 0$ for $t_s = F_p/(v_s - v_p)$ and $\pi_b = 0$ for $t_p = (F_b - F_p)/(v_p - v_b)$. This holds true independently of how many generator firms that are participating in the market. Costs are recovered in optimum whether x_i represents the aggregated capacity of generators owned by different firms, or it represents the capacity of one firm being the only owner of a given generator type *i*.

In a long-term market equilibrium, no individual firm has an incentive to change its capacity, given that the other firm's capacities (and all parameters) are kept constant. Under perfect market conditions with no barriers to exit and entry, no single firm f can influence the duration of the different generators (or load shedding) by changing its capacity. Profit maximization of firm f owning a peaker plant yields:

$$\max_{x_{f,p}} \pi_{f,p} = \int_{0}^{t_p} (p(t) - v_p) q_{f,p}(t) dt - F_p x_{f,p} = \left(\left(v_s - v_p \right) t_s - F_p \right) x_{f,p}$$
(32)

Since t_s is not influenced by the investment $x_{f,p}$ of firm f, the first-order optimality condition becomes:

$$\frac{\partial \pi_{f,p}}{\partial x_{f,p}} = 0 \Rightarrow (v_s - v_p)t_s - F_p = 0 \Rightarrow t_s = \frac{F_p}{(v_s - v_p)}$$
(33)

which exemplifies the well-known general result that the profit of each single generator company maximizes its profit in system optimum in a perfect market. Similarly, the market solution for the baseplant technology becomes:

$$\max_{x_{f,b}} \pi_{f,b} = \int_{0}^{T} (p(t) - v_b) q_{f,b}(t) dt - F_b x_{f,b}$$

$$= \left((v_s - v_b) t_s + (v_p - v_b) (t_p - t_s) - F_b \right) x_{f,b}$$

$$\frac{\partial \pi_{f,b}}{\partial x_{f,b}} = 0 \Rightarrow t_p = \frac{F_b - F_p}{(v_p - v_b)}$$
(34)
(34)
(35)

4 Variable Renewable Energy (VRE) power plants

VRE plants are powered by weather-driven and variable primary energy sources with zero fuel costs. Dependent on the conversion technology and control systems, the output of VRE plants can be controlled in both downwards (by reducing its output) or upwards (by initially running at a set-point lower than maximum possible) directions (Milligan et al. 2015). In this paper, we simplify the representation of VRE plants by assuming zero variable costs and full downward dispatch capability. By assuming that the VRE output scales linearly with its installed capacity, the relative power output variations become independent of the capacity of the VRE generator, and we can write $q_v(t) = AF_v(t) \cdot x_v$ where $AF_v(t)$ is the availability factor at time instant t. The assumption of linear scaling of VRE output is a significant simplification since e.g. different wind power sites have different wind conditions which lead to smoothing of aggregated output as a function of capacity. Nevertheless it is a common approximation in more detailed modelling work than what is presented here (Sepulveda et al. 2018; De Vita, Kielichowska, and Mandatowa 2018; Cole et al. 2016; de Sisternes, Jenkins, and Botterud 2016).

With zero marginal costs, VRE plants will always be dispatched first, and the conventional generators must cover the net demand

$$q_{nd}(t) = q_d(t) - \overline{q}_v(t) = q_d(t) - AF_v(t)x_v$$
(36)

where $\overline{q}_{v}(t)$ refer to the uncurtailed available VRE generation. The time parameter t now refers to the duration of the net demand, sorted from highest to lowest value over the year, as illustrated in Figure 2.



Figure 2. Load duration curve for demand (grey line) and net demand (black) line for a system with two conventional generators ("peak" and "base") and one VRE plant. The optimal capacities of peaker and baseplant with and without VRE are indicated in the figure as "new" and "old", respectively.

With a VRE plant added to the system in addition to the peaker and baseplant, the optimization problem becomes

$$\min_{x_i, q_k(t)} C = \sum_i F_i x_i + \sum_j v_j \int_0^T q_j(t) dt$$
(37)

s.t.
$$q_d(t) - \sum_k q_k(t) = 0$$
 (38)
 $-q_k(t) \le 0$

$$-q_k(t) \le 0 \tag{39}$$

$$q_l(t) - x_l \le 0 \tag{40}$$

$$q_{\nu}(t) - AF_{\nu}(t)x_{\nu} \le 0 \tag{41}$$

Sets:
$$i \in \{p, b, v\}$$
, $j \in \{s, p, b\}$, $k \in \{s, p, b, v\}$, $l \in \{p, b\}$

4.1 System optimality conditions

As for the case with only conventional generators, we first derive the conditions for optimal dispatch with given plant capacities. The Lagrangian of the operation problem for an arbitrary time instant t is:

$$\mathcal{L}_{op}(t) = \sum_{j} v_{j} q_{j}(t) + \lambda_{d}(t) \left(q_{d}(t) - \sum_{k} q_{k}(t) \right) + \sum_{l} \mu_{l}^{max}(q_{l}(t) - x_{l}) + \sum_{k} \mu_{v}^{max}(q_{v}(t) - AF_{v}(t)x_{v}) + \sum_{k} \mu_{k}^{min}(-q_{k}(t)) + \sum_{k} \mu_{k}^{min}($$

The KKT-conditions for this problem consist of (38)-(41), in addition to the following equations:

$$v_{l} - \lambda_{d}(t) + \mu_{l}^{max} - \mu_{l}^{min} = 0 \,\forall \, l \in \{p, b\}$$
(43)

$$-\lambda_d(t) + \mu_v^{max} - \mu_v^{min} = 0 \tag{44}$$

$$v_s - \lambda_d(t) - \mu_s^{min} = 0 \tag{45}$$

$$\mu_l^{max} \cdot (q_l(t) - x_l) = 0 \,\forall \, l \in \{p, b\}$$
(46)

$$\mu_v^{max} \cdot (q_v(t) - AF_v(t)x_v) = 0$$

$$\mu_k^{\min} \cdot \left(-q_k(t)\right) = 0 \ \forall \ k \in \{s, p, b, v\}$$

$$\tag{47}$$

From the KKT-conditions, we get the dispatch according to the merit-order similarly to Chapter 3, but now with the VRE plant added. The resulting optimal dispatch levels of each generator in each period are provided in Table 2, where the time parameter t is sorted after the net demand.

Table 2. System marginal cost and optimal operation for different net demand levels.

$t \leq t_s$:	$t_s < t \le t_p$:	$t_p < t \le t_b$:	$t > t_b$:
$x_b + x_p \le q_{nd}(t)$	$x_b \le q_{nd}(t) < x_b + x_p$	$0 \le q_{nd}(t) < x_b$	$q_{nd}(t) < 0$
$\lambda_d(t) = v_s$	$\lambda_d(t) = v_p$	$\lambda_d(t) = v_b$	$\lambda_d(t) = 0$
$q_v(t) = AF_v(t)x_v$	$q_{v}(t) = AF_{v}(t)x_{v}$	$q_{v}(t) = AF_{v}(t)x_{v}$	$q_v(t) = q_d(t)$
$q_b(t) = x_b$	$q_b(t) = x_b$	$q_b(t) = q_{nd}(t)$	$q_b(t) = 0$
$q_p(t) = x_p$	$q_p(t) = q_{nd}(t) - x_b$	$q_p(t) = 0$	$q_p(t) = 0$
$q_s(t) = q_{nd}(t) - x_p - x_b$	$q_s(t) = 0$	$q_s(t) = 0$	$q_s(t) = 0$

Following the method from Chapter 3, we can use the result in Table 2 to express the capacity investment problem as:

$$\min_{x_p, x_b, x_v} C = F_p x_p + F_b x_b + F_v x_v + v_s \int_0^{t_s} [q_d(t) - AF_v(t)x_v - x_p - x_b] dt + v_p \cdot \left(\int_0^{t_s} x_p dt + \int_{t_s}^{t_p} [q_d(t) - CF_v(t)x_v - x_b] dt \right) + v_b \cdot \left(\int_0^{t_p} x_b dt + \int_{t_p}^{t_b} (q_d(t) - AF_v(t)x_v) dt \right)$$
(48)

which is equivalent to problem (37)-(41) under the condition that all generators are present in the system. From the resource model of the VRE plant, we can calculate the average availability factor $AF_v^{[t_1,t_2]}$ between two time steps t_1 and t_2 , and use this to solve the integral

$$\int_{t_1}^{t_2} AF_v(t) dt = (t_2 - t_1) \cdot AF_v^{[t_1, t_2]}$$
(49)

Inserting into (49):

$$\min_{x_{p}, x_{b}, x_{v}} C = F_{p} x_{p} + F_{b} x_{b} + F_{v} x_{v} + v_{s} \left(E_{d}^{[0,t_{s}]} - \left(x_{p} + x_{b} + AF_{v}^{[0,t_{s}]} x_{v} \right) t_{s} \right) + v_{p} \left(E_{nd}^{[t_{s},t_{p}]} + \left(x_{p} + x_{b} + AF_{v}^{[t_{s},t_{p}]} \right) t_{s} - \left(x_{b} + AF_{v}^{[t_{s},t_{p}]} x_{v} \right) t_{p} \right) + v_{b} \left(E_{nd}^{[t_{p},t_{b}]} + \left(x_{b} + AF_{v}^{[t_{p},t_{b}]} x_{v} \right) t_{p} - AF_{v}^{[t_{p},t_{b}]} x_{v} t_{b} \right)$$
(50)

where we also have inserted the total energy demand $E_d^{[t_1,t_2]}$ over the period $[t_1, t_2]$. The first-order optimality conditions of (50) with respect to x_p and x_b becomes identical to the case without VRE given by (27) and (28). The optimal capacities are now found from the net demand curve, which is a function of the VRE capacity. Derivation of the cost function (50) with respect to x_v yields:

$$\frac{\partial C}{\partial x_{v}} = F_{v} - v_{s}AF_{v}^{[0,t_{s}]}t_{s} - v_{p}AF_{v}^{[t_{s},t_{p}]}(t_{p} - t_{s}) - v_{b}AF_{v}^{[t_{p},t_{b}]}(t_{b} - t_{p})$$
(51)

$$\frac{\partial C}{\partial x_{\nu}} = 0 \Longrightarrow F_{\nu} = \nu_s A F_{\nu}^{[0,t_s]} t_s + \nu_p A F_{\nu}^{[t_s,t_p]} (t_p - t_s) + \nu_b A F_{\nu}^{[t_p,t_b]} (t_b - t_p)$$
(52)

During all periods when the net demand is higher than zero, each additional kWh from the VRE plant will reduce the output of the marginal generator with the same amount, while all other generator outputs are unchanged. In optimum, the marginal value of this replacement of thermal generators and load shedding will equal the marginal cost of installing the corresponding capacity. Since t_s and t_p are known, we can solve (52) with respect to t_b :

$$t_b = t_p + \left(v_b A F_v^{[t_p, t_b]}\right)^{-1} \cdot \left(F_v - v_s t_s A F_v^{[0, t_s]} - v_p (t_p - t_s) A F_v^{[t_s, t_p]}\right)$$
(53)

A special case occurs when the optimal x_v is non-zero, but still not high enough to cause surplus generation, i.e. the optimal t_b is equal to T. The optimality condition (52) then indirectly determines x_v through the availability factor AF_v . Consider a situation where $x_v = 0$, and the right-hand side of (52) is larger than the left-hand side, i.e. it is profitable marginally increase the VRE capacity. A marginal increase in VRE capacity pushes the net demand curve downwards, and thus also changes the average availability factor in the different price segments. As more VRE capacity is installed, more of the generation is pushed towards the periods of the year with less net demand and lower prices. At a certain value of x_v , the marginal benefit for the system equals the marginal investment cost F_v . This can occur at $t_b = T$ or $t_b < T$, depending on the VRE investment cost. Figure 15 in Appendix A exemplifies this result for a case using European offshore wind data.

Thus, under the assumption of one type of VRE plant which scales linearly, it is the duration of the thermal generator with lowest marginal cost (in our case t_b) and the VRE output variations, that determines its optimal capacity x_v . The duration t_b expresses the duration of the year when the net demand is non-negative. The optimal VRE capacity x_v is the capacity that leads to net demand of zero at exactly t_b , and can easily be found from the load and VRE resource data. When x_v is known, the peaker and baseplant capacities can be found directly from the net demand curve.

In Figure 2, the base generator capacity is reduced significantly while the peak generator capacity is marginally increased when VRE is included, which is a result of the illustrated VRE variability; higher availability factor during low loads than high loads. Maximum load shedding is therefore higher with

VRE, to ensure optimal duration of the peaker with the changed net demand. However, note that the duration of load shedding t_s remain unchanged with the introduction of VRE. If the energy cost of the VRE plant is sufficiently low, it might push one of the thermal generators out of the solution space. To find the true optimum of the capacity investment problem (37)-(41), we must therefore compare the result of sub-problem (48) with the remaining options, i.e. VRE+peaker and VRE+baseplant. With only one thermal generator, here denoted *i*, the problem becomes a simpler variant of (48):

$$\min_{x_i, x_v} C = F_i x_i + F_v x_v + v_s \left(E_d^{[0, t_s]} - \left(x_i + A F_v^{[0, t_s]} x_v \right) t_s \right) \\ + v_i \left(E_{nd}^{[t_s, t_i]} + \left(x_i + A F_v^{[t_s, t_i]} x_v \right) t_s - A F_v^{[t_s, t_i]} x_v t_i \right)$$
(54)

with the VRE optimality condition:

$$t_{i} = t_{s} + \frac{\left(F_{v} - v_{s}t_{s}AF_{v}^{[0,t_{s}]}\right)}{v_{i}AF_{v}^{[t_{s},t_{i}]}}$$
(55)

Again, the optimal VRE capacity is found from the net demand curve that fulfils $q_{nd}(t_i) = 0$.

4.2 Profit maximization and cost recovery

Given a market where any number of VRE plants of the same type v is fully exposed to the electricity market prices, their aggregated profit function is:

$$\pi_{v} = AR_{v} - AC_{v} = \int_{0}^{T} p(t)q_{v}(t)dt - F_{v} \cdot x_{v} = x_{v} \int_{0}^{T} p(t)AF_{v}(t)dt - F_{v} \cdot x_{v}$$
(56)

In a market based on marginal cost pricing, we have from Table 2:

$$\pi_{v} = x_{v} \left(v_{s} \int_{0}^{t_{s}} AF_{v}(t) dt + v_{p} \int_{t_{s}}^{t_{p}} AF_{v}(t) dt + v_{b} \int_{t_{p}}^{t_{b}} AF_{v}(t) dt \right) - F_{v} \cdot x_{v}$$

$$\pi_{v} = \left(v_{s} AF_{v}^{[0,t_{s}]} t_{s} + v_{p} AF_{v}^{[t_{s},t_{p}]}(t_{p} - t_{s}) + v_{b} AF_{v}^{[t_{p},t_{b}]}(t_{b} - t_{p}) - F_{v} \right) \cdot x_{v}$$
(57)

We see from (57) that the VRE power plants exactly recover their cost in a market based on marginal cost pricing if plant capacities are according to system optimum: $\pi_v = 0$ leads to (52) and the corresponding optimal duration of the baseplant in (53).

Maximization of the profit function (56) for a firm f owning some VRE capacity $x_{v,f}$ gives the same result:

$$\frac{\delta \pi_{v,f}}{\delta x_{v,f}} = 0 \Rightarrow F_v = \int_0^T p(t) A F_v(t) dt$$
(58)

where we have required that each owner of the VRE plant is too small to change the generator durations alone. Thus, market based VRE investments reaches an equilibrium at system optimum where it is not profitable for any single owner to change its capacity given that all external parameters remain constant.

5 Electric Energy storage (EES)

EES are applied for numerous services in the power system, such as load shifting, energy arbitrage, operating reserves, backup capacity, transmission and distribution deferral, among others (Dell and Rand 2001; Denholm et al. 2010; Akhil et al. 2015). Conventional storage technologies like pumped hydro has been used for shifting energy from base load to peak load periods for decades (Mcdaniel and Gabrielle 1966; Botterud, Levin, and Koritarov 2014; Harby et al. 2013). Newer storage technologies like Li-Ion batteries and flow batteries has received a lot of attention the last two decades, both with respect to renewable energy integration, price arbitrage and grid services (Aneke and Wang 2016; Denholm et al. 2010; Akhil et al. 2015). In this paper we will keep EES modelling at a simple and general level, in order to derive analytical expressions for all optimality conditions when EES is present in the system. A similar load duration method for analysis of storage systems is reported in (Steffen and Weber 2013), which derives optimality conditions for EES, but is limited to cases where the VRE plant capacity is an external input and not part of the system optimum.

Inclusion of EES in a modelling framework based on duration curves cannot be achieved without major simplifications, due to the chronological nature of the EES dispatch. EES is limited by the power capacity, energy capacity and round-trip efficiency of the storage devices. Capacity degradation mechanisms and other aging effects are also important to consider for certain EES types such as most electrochemical batteries (Xu et al. 2018). The aim of the modelling framework presented here is to supplement, not replace detailed optimization studies. To get a principal understanding of the potential for EES under ideal conditions, we therefore choose a representation of storage which takes explicitly into account the power capacity and round-trip efficiency, but only indirectly accounts for the energy capacity. This is obviously a simplistic representation, but, as will be shown, is still useful to indicate what potential role ESS have in power market equilibrium.

5.1 EES with conventional generators

5.1.1 System optimality conditions

We will first investigate how the optimality conditions for thermal generators are influenced by the presence of ESS. The output of the ESS will depend on the installed capacity of conventional generators when we use ESS for load shifting. Hence, it follows that the optimal duration of conventional generators also will change in the presence of storage, as earlier derived in (Steffen and Weber 2013). We simplify the analysis by first assuming that we can shift energy from any instance of time to any other, i.e. that the storage only has a power (kW) constraint, as also assumed in (Steffen and Weber 2013). The optimal operational strategy of an EES unit will then be to shift as much energy as possible to the hours with load shedding. If EES charged by baseload is cheaper than the peaking generator, the ESS will also be discharged when the peaker is the marginal unit, as illustrated in Figure 3.



Figure 3. Load duration curve including energy storage, which is charged by the base plant. The storage device sets the price between t_p and t_e . The grey areas are the charging and discharging energy.

Referring to Figure 3, we can split the whole year into four main periods:

 $[0, t_s]$: EES is discharging at full power during load shedding. A marginal increase in x_e causes a marginal decrease of load shedding over the whole period. From energy storage preservation it follows that this generation must be compensated by charging of the same amount divided by the round-trip efficiency η_e . We have assumed that the energy storage capacity is sufficiently large to be charged only by the cheapest generator.

 $\langle t_s, t_p]$: EES is discharging at full power when the peaker is the marginal generator. A marginal increase in x_e causes a marginal decrease in generation from the peaker. As above, this is compensated by marginally more charging energy. Again, this operating strategy assumes that the EES energy capacity is sufficiently large and that $v_p > \eta_e v_b$.

 $\langle t_p, t_e]$: EES is the marginal generator. A marginal increase in demand increases marginally the discharge by the same amount. This leads to a marginal increase in charging sometime during $\langle t_e, T]$ of the same amount, divided by the round-trip efficiency η_e .

 $\langle t_e, T]$: The baseplant is the marginal generator, while the EES is either charging or stand-by.

The system cost minimization problem is identical to the general problem described in Chapter 2 without the VRE plant:

$$\min_{x_i} C = \sum_i F_i x_i + \sum_j v_j \int_0^T q_j(t) dt$$
(59)

s.t.
$$q_d(t) - \sum_k q_k(t) + q_{e^-}(t) = 0$$
 (60)

$$-q_k(t) \le 0$$
, $-q_{e^-}(t) \le 0$ (61)

$$q_i(t) - x_i \le 0$$
, $q_{e-}(t) - x_e \le 0$ (62)

$$\eta_{e} \int_{0}^{T} q_{e-}(t)dt - \int_{0}^{T} q_{e}(t)dt = 0$$
(63)

Sets:
$$i \in \{p, b, e\}, j \in \{s, p, b\}, k \in \{s, p, b, e\}$$

We first consider the optimal operation during the whole discharging period $[0, t_e]$. According to the assumptions stated above, all generated power from the EES comes from the baseplant with variable cost v_b . With a storage round-trip efficiency of η_e , the variable cost of generation from EES is $v_e = v_b/\eta_e$. Under the assumption that there is enough energy available for charging the EES, it can be treated as a conventional generator with variable cost v_b/η_e during the discharge period. Based on the operation problem formulation (15)-(18) with corresponding Lagrangian (19) and KKT-conditions (20)-(23), which applies for any number of conventional generators, we derive the optimal operating policy during discharging as shown in Table 3. In the charging period $\langle t_e, T]$, the baseplant supplies the consumer demand and the charging power. Since the marginal cost of generation is constant equal to v_b over the whole period $\langle t_e, T]$, the charging can take place any time in this interval as long as the storage conservation (63) is fulfilled. The cost of charging is thus equal to v_b .

Table 3. System marginal cost and optimal operation for different demand levels with EES charged by the baseplant, under the assumption that $v_p > \eta_e v_b$.

$t \leq t_s$:	$t_s < t \le t_p$:	$t_p < t \le t_e$:	$t > t_e$:
$x_b + x_e + x_p \le q_d(t)$	$0 \le q_d(t) - x_b - x_e < x_p$	$0 \le q_d(t) - x_b < x_e$	$q_d(t) + q_{e-}(t) > x_b$
$\lambda_d(t) = v_s$	$\lambda_d(t) = v_p$	$\lambda_d(t) = v_b/\eta_e$	$\lambda_d(t) = v_b$
$q_b(t) = x_b$	$q_b(t) = x_b$	$q_b(t) = x_b$	$q_b(t)$
			$= q_d(t) + q_{e-}(t)$
$q_e(t) = x_e$	$q_e(t) = x_e$	$q_e(t) = q_d(t) - x_b$	$q_e(t) = 0$
$q_p(t) = x_p$	$q_p(t) = q_d(t) - x_b - x_e$	$q_p(t) = 0$	$q_p(t) = 0$
$q_s(t) = q_d(t) - x_p - x_e - x_b$	$q_s(t) = 0$	$q_s(t) = 0$	$q_s(t) = 0$

Table 3 contains the the operational constraints (60)-(62) for problem (59) solved for optimal operation over the whole duration [0, T]. We can use these operational conditions to express the total cost minimization problem as:

$$\min_{x_p, x_b, x_e} C = F_p x_p + F_b x_b + F_e x_e + v_s \int_0^{t_s} [q_d(t) - x_p - x_e - x_b] dt + v_p \cdot \left(\int_0^{t_s} x_p \, dt + \int_{t_s}^{t_p} [q_d(t) - x_b - x_e] \, dt \right) + v_b \cdot \left(\int_0^{t_e} x_b \, dt + \int_{t_e}^{T} [q_d(t) + q_{e^-}(t)] \, dt \right)$$
s.t. $\eta_e \int_{t_e}^{T} q_{e^-}(t) dt - \int_0^{t_e} q_e(t) dt = 0$
(64)

By the use of the storage constraint (65) and the optimal operation strategy from Table 3, the last term of the objective function can be substituted by:

$$\int_{t_e}^{T} q_{e^-}(t)dt = \eta_e^{-1} \int_0^{t_e} q_e(t)dt = \eta_e^{-1} \left(\int_0^{t_p} x_e \, dt + \int_{t_p}^{t_e} [q_d(t) - x_b] \, dt \right)$$
(66)

The optimization problem can now be written as

$$\min_{x_p, x_b, x_e} C = F_p x_p + F_b x_b + F_e x_e + v_s \int_0^{t_s} [q_d(t) - x_p - x_e - x_b] dt + v_p \cdot \left(\int_0^{t_s} x_p \, dt + \int_{t_s}^{t_p} [q_d(t) - x_b - x_e] \, dt \right) + \eta_e^{-1} v_b \cdot \left(\int_0^{t_p} x_e \, dt + \int_{t_p}^{t_e} [q_d(t) - x_b] \, dt \right) +$$

$$v_b \cdot \left(\int_0^{t_e} x_b \, dt + \int_{t_e}^{T} q_d(t) \, dt \right)$$
(67)

which is identical to a three-generator problem where the mid-merit generator has marginal $\cot \eta_e^{-1} v_b$. The optimality condition for the peaker is unchanged from (27), while the EES and baseplant conditions becomes:

$$\frac{\partial C}{\partial x_e} = 0 \Longrightarrow F_e - (v_s - \eta_e^{-1} v_b) t_s - (v_p - \eta_e^{-1} v_b) (t_p - t_s) = 0$$
⁽⁶⁸⁾

$$\implies t_p = \frac{F_e - F_p}{v_n - \eta_e^{-1} v_b} \tag{69}$$

$$\frac{\partial C}{\partial x_b} = 0 \Longrightarrow F_b - (v_s - v_b)t_s - (v_p - v_b)(t_p - t_s) - (\eta_e^{-1}v_b - v_b)(t_e - t_p)$$
(70)

$$\implies t_e = \frac{F_b - F_e}{v_b(\eta_e^{-1} - 1)} \tag{71}$$

These conditions apply for an ideal storage device which can move energy from any instant of time to another, only limited by its power capacity and round-trip efficiency, i.e. with unbounded energy capacity. For real-world applications, the last assumption is of course too optimistic, so the results should be regarded as an upper limit to how much storage can contribute with in system optimum. Furthermore, the durations above only apply for storage efficiencies less than 1, which is always the case in real systems. The conversion losses make discharge more expensive in operation than direct supply from the generator used for charging. From (71) we see that the optimality condition leads to an infeasible solution for round-trip efficiency of 1. The discharge duration according to (74) goes to infinity; you will replace as much thermal energy as you can with ESS as long as the fixed costs of ESS are lower than for the baseload plant. However, the discharge duration is strictly limited by the storage preservation (65) which always applies, since it is not possible to discharge more energy than available from the charging period.

The optimality conditions state that the optimal duration of the peaker is proportional to the EES fixed $\cos F_e$, while the baseplant duration is negatively proportional to F_e . The durations are equal when

$$F'_{e} = \left(F_{b} - F_{p}\right) \frac{\left(v_{p} - \eta_{e}^{-1}v_{b}\right)}{\left(v_{p} - v_{b}\right)}$$
(72)

 $F_e > F'_e$ implies that $t_p > t_e$, which is not feasible if $v_p > \eta_e v_b$, i.e. EES investment is zero in that case. If $v_p < \eta_e v_b$, on the other hand, the EES and the peaker will shift places in the merit-order, and we can directly set up the optimality conditions in a similar manner as above::

$$\frac{\partial C}{\partial x_e} = 0 \Longrightarrow t_s = \frac{F_e}{v_s - \eta_e^{-1} v_b}$$
(73)

$$\frac{\partial C}{\partial x_p} = 0 \Longrightarrow t_e = \frac{F_p - F_e}{v_p (\eta_e^{-1} - 1)}$$
(74)

where it is still assumed that all stored energy is generated from the baseplant. For a less optimistic case where EES is charged only by the peaker, it is necessary to replace v_b in the denominator of (73) with v_p . The optimality condition for the baseplant is given by (28), since the peaker again is adjacent to the baseplant in the merit-order.

5.1.2 Cost recovery

The profit function of EES is

$$\pi_{e} = AR_{e} - AC_{e} = \int_{0}^{T} p(t) \left(q_{e}(t) - q_{e-}(t) \right) dt - F_{e} \cdot x_{e}$$
(75)

where the instantaneous charging and discharging power is given by the storage operation strategy and is generally a function of the storage capacity (power and energy) and the market price. In our modelling framework, we take explicitly into account the power constraint by (61) and (62), while energy storage conservation is fulfilled on an annual basis by (63). In the previous section, it was shown that the marginal cost of energy generated from the EES is equal to the variable cost of the marginal thermal generator that is used for charging, divided by the round-trip efficiency.

By using the segments from the KKT-conditions in Table 3 and marginal cost pricing, the profit function can be expressed as

$$\pi_{e} = v_{s} x_{e} t_{s} + v_{p} x_{e} (t_{p} - t_{s}) + p_{e} \int_{t_{p}}^{t_{e}} q_{e}(t) dt - v_{b} \int_{t_{e}}^{T} q_{e-}(t) dt - F_{e} \cdot x_{e}$$
(76)

where p_e is the short-term electricity price during period $[t_p, t_e]$, when the storage is the marginal generator. From the KKT-conditions, we know that this price should be equal to $\eta_e^{-1}v_b$ to ensure optimal operation, since this is the marginal cost of charging the EES. In other words, we assume perfect knowledge of the future to determine the value of stored energy, by applying a deterministic model of the system on an annual basis. In a real market setting, on the other hand, the future prices, demand, weather conditions etc. are not known, and the marginal value of the stored energy, i.e. the opportunity cost (Førsund 2015) will consequently change over time. In hydropower planning this is referred to as the "water value" (Stage and Larsson 1961; Fosso et al. 1999; Førsund 2015), which is typically calculated as an expectation value based on stochastic representation of future unknown parameters. For detailed treatment of uncertainty in EES operation strategies, see (Xu, Botterud, and Korpås 2019). The use of stochastic parameters for storage valuation in the duration curve framework is an area for future research.

By applying $p_e = \eta_e^{-1} v_b$, and the storage conservation $\eta_e \int_{t_e}^{T} q_{e-}(t) dt = x_e t_p + \int_{t_p}^{t_e} q_e(t) dt$, the profit function becomes:

$$\pi_e = v_s x_e t_s + v_p x_e (t_p - t_s) + \eta_e^{-1} v_b x_e t_p - F_e \cdot x_e$$
(77)

By setting $\pi_e = 0$ we obtain the optimality condition for EES given by (69). In perfect competition where no EES owner is large enough to change the durations t_s or t_p , profit maximization yields

$$\frac{\partial \pi_e}{\partial x_e} = 0 \Longrightarrow v_s t_s + v_p (t_p - t_s) + \eta_e^{-1} v_b t_p - F_e = 0$$
(78)

which also gives the optimality condition (69). As shown above, the optimal price for discharging energy is equal to the marginal cost of charging, under the assumption of perfect foresight and annual storage balance.

5.2 Energy storage in systems with VRE

Building on the storage model for conventional generators from Section 5.1, we explore two main ESS operating strategies when introducing VRE. The first principle is to use storage exclusively for utilizing surplus VRE, which is discussed further in Section 5.2.1 and 5.2.2. The second operating principle is general price arbitrage regardless of the source of energy, which is covered in Section 5.2.3.

In the storage model introduced in the previous section, it was assumed that it was possible to move energy from any instant of time to another over the course of the year. Although very optimistic, the assumption can at least partly be justified for systems with only conventional generators, which alternate in setting the price on a daily or weekly basis. With VRE, this assumption is more questionable, as the time of surplus VRE generation and the time of peak load might be in different seasons of the year and would require very large amounts of storage capacity. Therefore, we also present a restrictive operational strategy in Section 5.2.2, where we assume that the stored VRE energy is only able to replace base plant generation due to storage limitations.

5.2.1 VRE storage with unlimited energy capacity

In the ideal case with no storage limits, it is possible to discharge the stored energy in the periods with highest price first. This is illustrated in Figure 4 for a system with one peaker p, one base plant b, one VRE plant p and one energy storage device e. In the illustrated case, there is enough surplus VRE energy to discharge during all periods where load shedding and conventional generators sets the price for electricity. A new variable t_v is introduced in the figure, which is defined as the time instant when the negative net load equals the charging capacity: $q_{nd}(t_v) = -x_e$.



Figure 4. Net load duration curve including energy storage charged by surplus VRE energy. Left: Storage as individual generator and consumer. Right: Storage as part of the net demand. These are equivalent representations.

Consider the illustrated case in Figure 4 where the available amount of discharged energy is limited by the power capacity of storage, the duration of VRE surplus and the storage efficiency. Since we assume unbounded energy storage, it is most beneficial to discharge at full power during load shedding $[0, t_s]$ and the peaker period $[t_s, t_p]$ and discharge the rest of the stored energy in the base plant period $[t_p, t_b]$. The available discharge energy during $[t_p, t_b]$ is limited by:

$$E_{e}^{[t_{p},t_{b}]} = \int_{t_{p}}^{t_{b}} q_{e}(t)dt = \eta_{e} \int_{t_{b}}^{T} q_{e-}(t)dt - x_{e}t_{p} = \eta_{e}E_{e-}(x_{v},x_{e}) - x_{e}t_{p}$$
(79)

where it is indicated that the annual charging energy is a function of both the VRE capacity and the storage capacity, as visualized in Figure 4. Equation (79) holds true if there is enough charging energy available to discharge at full power during the whole period $[0, t_p]$.

It is indifferent at which time during $[t_p, t_b]$ that the storage is discharged, since the base plant is the marginal generator over the whole period. Note that it is not possible to cover a marginal load increase in $[t_p, t_b]$ by stored energy, as all the surplus energy from $[t_b, T]$ is already utilized. Thus, the marginal generator in $[t_p, t_b]$ is the base plant, even with storage present, and the short-term marginal cost in this time segment is v_b . However, in the long-term, there exists two additional options to cover a marginal demand increase during $[t_p, t_b]$:

- Increasing the power capacity of the storage x_e while the VRE capacity is constant. Referring to Figure 4, this option increases the amount of charged energy during $[t_b, T]$.
- Increasing the VRE capacity x_v while x_e is constant. This option moves the net demand curve to the left, and thus increases the length of $[t_b, T]$ which leads to more available stored energy.

As will be shown in the next section, the optimal response to a marginal increase in demand is a combination of these two options, i.e. increasing both x_e and x_v , according to their optimality conditions.

5.2.1.1 Minimization of system costs

The system cost minimization problem is identical to the main problem (9)-(14), except that we add the charging constraint:

$$q_{e-}(t) - q_{nd}(t) \le 0$$
(80)

which ensures that only surplus VRE energy is stored (general price arbitrage is treated in Section 5.2.3). We first find the optimal operation in each of the periods indicated in Figure 4, under the assumption that there is sufficient stored energy available to discharge at full power during the whole period $[0, t_p]$:

 $[0, t_s]$: All dispatchable generators are operated at full power, so the marginal cost is given by load shedding. Hence $\lambda_d = v_s$ as before.

 $\langle t_s, t_p]$: The baseplant and EES are operated at full power, and the marginal plant is the peaker. Hence $\lambda_d = v_p$ as before.

 $\langle t_p, t_b \rangle$: The demand is covered by the baseplant, VRE and EES. The baseplant is the marginal generator in the whole period, hence $\lambda_d = v_b$. Consider that the EES covers a marginal demand increase over a small time interval Δt during $[t_p, t_b]$: $\Delta P_d \cdot \Delta t = \Delta P_e \cdot \Delta t$ The marginal increase in EES generation will reduce the storage level according to the instantaneous storage balance $\Delta E_{stored} = \eta_e^{-1}\Delta P_e \cdot \Delta t = \eta_e^{-1}\Delta P_d \cdot \Delta t$. However, the annual storage balance (14) must be fulfilled. Since all available VRE energy is already utilized, the EES is forced to charge energy from the baseplant to cover the storage deficit caused by the marginal demand increase: $\Delta P_b \cdot \Delta t = \eta_e^{-1}\Delta P_d \cdot \Delta P_d$. Hence, without new investments, a marginal increase in EES generation during $[t_p, t_b]$ must be covered by the baseplant. Due to the storage efficiency, $\eta_e < 1$, it will always be cheaper to cover a marginal demand increase during this period directly by the baseplant, i.e. the baseplant is the marginal generator and $\lambda_d = v_b$ over the whole period.

 $\langle t_b, t_v \rangle$: In this period, there is surplus of VRE energy, and the EES is charged below its full capacity, $0 \le q_{nd}(t) < x_e$. Hence, during this period, the EES is the marginal load, and the marginal cost of electricity is set by the marginal value of stored energy, $\lambda_d = \eta_e v_b$. This corresponds to the increase in required baseplant generation in period $\langle t_p, t_b \rangle$ if EES charging is reduced in period $\langle t_b, t_v \rangle$, accounting for the EES efficiency. During $[t_b, t_v]$, there are two competing options for supplying a marginal increase in demand: 1) Starting up the baseplant with marginal cost v_b . 2) Reducing the charging of EES with the same amount $\Delta P_{e^-} \cdot \Delta t = -\Delta P_d \cdot \Delta t$. This gives less stored energy $\Delta E_{stored} = -\eta_{e,ch}\Delta P_d \cdot \Delta t$, which reduces the available energy for discharging: $\Delta E_e^{[t_p, t_b]} = \eta_{e,dch}\Delta E_{stored} = -\eta_e\Delta P_d \cdot \Delta t$. The reduced energy from discharging must be substituted by the marginal generator in the least expensive discharging period, i.e. the baseplant: $\Delta P_b = -\frac{\Delta E_e^{[t_p, t_b]}}{\Delta t} = \eta_e \Delta P_d$, with marginal cost v_b . Hence, the cost of a marginal increase in demand ΔP_d is $\lambda_d = \eta_e v_b$, which is always cheaper than the baseplant since $\eta_e < 1$.

 $\langle t_{\nu}, T \rangle$: There is surplus of VRE energy, and the EES is charged at full power. Hence, VRE is the marginal plant and $\lambda_d = 0$.

With the optimal operating strategies for each period described above, we can formulate the total cost minimization problem:

$$\min_{i} C = \sum_{i} F_{i} x_{i} + v_{s} \int_{0}^{t_{s}} [q_{d}(t) - AF_{v}(t)x_{v} - \sum_{j} x_{j}] dt + v_{p} \cdot \left(\int_{0}^{t_{s}} x_{p} dt + \int_{t_{s}}^{t_{p}} [q_{d}(t) - AF_{v}(t)x_{v} - \sum_{k} x_{k}] dt\right) + v_{b} \cdot \left(\int_{0}^{t_{p}} x_{b} dt + \int_{t_{p}}^{t_{b}} [q_{d}(t) - AF_{v}(t)x_{v} - (81) q_{e}(t)] dt\right) \forall i \in \{p, b, e, v\}, j \in \{p, b, e\}, k \in \{e, b\}$$
s.t. $\eta_{e} \int_{t_{e}}^{T} q_{e-}(t) dt - \int_{0}^{t_{e}} q_{e}(t) dt = 0$
(82)

For this problem, the storage constraint (82) is equivalent to

$$E_{e}^{[t_{p},t_{b}]} = \eta_{e} E_{e^{-}}^{[t_{b},t_{v}]} + \eta_{e} x_{e} (T - t_{v}) - x_{e} t_{p}$$
(83)

where we have used the result that the EES is charging at maximum capacity during $\langle t_{\nu}, T \rangle$ and discharging at maximum capacity during $[0, t_p]$. Substituting (83) into (81):

$$\min_{i} C = \sum_{i} F_{i} x_{i} + v_{s} E_{d}^{[0,t_{s}]} + v_{p} E_{d}^{[t_{s},t_{p}]} + v_{b} E_{d}^{[t_{p},t_{b}]} - (x_{p} + x_{b} + x_{e})(v_{s} - v_{p})t_{s} - (x_{b} + x_{e})(v_{p} - v_{b})t_{p} - x_{e}\eta_{e}v_{b}(T - t_{v}) - \eta_{e}v_{b}E_{e^{-}}^{[t_{b},t_{v}]} - x_{v}\left(v_{s}CF_{v}^{[0,t_{s}]}t_{s} + v_{p}CF_{v}^{[t_{s},t_{p}]}(t_{p} - t_{s}) + v_{b}CF_{v}^{[t_{p},t_{b}]}(t_{b} - t_{p})\right)\forall i \in \{p, b, e, v\}$$
(84)

Problem (84) without additional constraints minimizes the system cost under the following assumptions:

- EES is only bounded by power capacity and yearly energy balance. EES operation is treated deterministic, i.e. with full foresight.
- EES is only used for storing surplus VRE energy that otherwise would have been curtailed.
- All power plants (peaker, baseload, EES and VRE) forms part of the solution. Other combinations form a simpler subset, and must be treated separately.
- It is enough stored energy available for the EES to generate at full power during the whole period $[0, t_p]$. Cases with less available charging are simpler variants of the problem and must be treated separately.

It is possible to derive the first-order optimality conditions for all plants directly from (79). For the peaker and the baseload, we obtain the same result as in Chapter 3 since these generators have not changed position in the merit order, see (27) and (28). For the EES plant, we get:

$$\frac{\partial C}{\partial x_e} = F_e - (v_s - v_p)t_s - (v_p - v_b)t_p - \eta_e v_b(T - t_v) - \eta_e v_b \frac{\partial E_{e^-}^{[t_b, t_v]}}{\partial x_e}$$
(85)

where the last term is zero, since the EES is the marginal load during $\langle t_b, t_b \rangle$ and therefore operates lower than its capacity. Setting $\frac{\partial C}{\partial x_e} = 0$, the optimality condition for the EES becomes:

$$\eta_e(T - t_v) = \frac{F_e - F_b}{v_b} \tag{86}$$

$$\Rightarrow t_{v} = T + \frac{F_{b} - F_{e}}{\eta_{e} v_{b}}$$
(87)

Equation (86) is equivalent to the optimality condition of a generator with zero variable cost, with a duration $t^* = \eta_e(T - t_v)$. The duration $\eta_e(T - t_v)$ expresses how long time it is necessary to charge the EES at full power, accounting for the storage losses η_e . Rearranging to (87), we get directly the unknown duration t_v , which is the time instant when the available surplus power is equal to the charging capacity x_e . An equivalent definition of t_v is the duration of full VRE energy utilization, i.e. the number of hours of the year without VRE curtailment. From (87) we see that t_v is proportional to $-F_e$, i.e. if the fixed cost of EES is decreased, less hours with zero price is needed to make the EES beneficial for the system. The threshold for EES to be economically viable is when $t_v = t_b$. If the calculated t_v is lower than t_b , this implies a negative charging capacity x_e which is infeasible (See Figure 4).

The baseplant duration t_b is found from the optimality condition for the VRE plant. From (84):

$$\frac{\partial C}{\partial x_{v}} = F_{v} - v_{s}AF_{v}^{[0,t_{s}]}t_{s} - v_{p}AF_{v}^{[t_{s},t_{p}]}(t_{p} - t_{s}) - v_{b}AF_{v}^{[t_{p},t_{b}]}(t_{b} - t_{p}) - \eta_{e}v_{b}\frac{\partial}{\partial x_{v}}E_{e^{-}}^{[t_{b},t_{v}]}$$
(88)

Previously in this section, we showed that a marginal increase in demand during $\langle t_b, t_v \rangle$ caused a marginal decrease in charging of the same amount since the EES is the marginal load unit in this period. If instead the VRE capacity increases marginally, the net load during the period will decrease by $AF_v^{[t_s,t_p]}(t_v - t_b)$. Since the EES does not operate at its maximum charging capacity during $\langle t_b, t_v \rangle$, it

can absorb the whole amount of additional VRE energy. Thus, the marginal increase in charging energy is $\frac{\partial}{\partial x_v} E_{e^-}^{[t_b, t_v]} = A F_v^{[t_b, t_v]}(t_v - t_b)$. The optimality condition for VRE becomes:

$$\frac{dC}{dx_v} = 0 \Longrightarrow F_v = v_s t_s \cdot AF_v^{[0,t_s]} + v_p (t_p - t_s) \cdot AF_v^{[t_s,t_p]} + v_b (t_b - t_p) \cdot AF_v^{[t_p,t_b]}$$

$$(89)$$

$$t_{b} = \frac{F_{v} - v_{s}t_{s} \cdot AF_{v}^{[0,t_{s}]} - v_{p}(t_{p} - t_{s}) \cdot AF_{v}^{[t_{s},t_{p}]} - v_{b}\left(\eta_{e}t_{v}AF_{v}^{[t_{b},t_{v}]} - AF_{v}^{[t_{p},t_{b}]}t_{p}\right)}{v_{b}\left(AF_{v}^{[t_{p},t_{b}]} - \eta_{e}AF_{v}^{[t_{b},t_{v}]}\right)}$$
(90)

We see that the duration of the baseload plant, which determines the optimality condition for VRE, is dependent of the characteristic of the EES. If we set $t_v = t_b$ (corresponding to zero EES investment) in (90), the duration t_b becomes equal to the result for the case without EES given in (53). We find from (90) that $\frac{\partial t_b}{\partial t_v} < 0$. This shows that an investment in EES (corresponding to $t_v > t_b$) leads to a higher investment of VRE in optimum, since a reduction in t_b only can be caused by increasing x_v as long as the demand is not changing.

When t_b and t_v has been calculated, it is straightforward to find the optimal VRE capacity and EES capacity from the net demand curve: $q_{nd}(t_b) = 0$ and $q_{nd}(t_v) = -x_e$.

5.2.1.2 Cost recovery of EES

As already mentioned, we present results for the case where no generator types are entirely pushed out of the market. Both the peaker and base plant will be present in the system. VRE and EES will cause a change in the capacity and generation of the base and peaker, but the duration of load shedding (optimality condition for the peaker) and duration of the peaker (optimality condition for the base plant) are both unchanged.

The profit function of the EES is set up similarly to Section 5.1.2:

$$\pi_{e} = \int_{0}^{T} p(t) (q_{e}(t) - q_{e-}(t)) dt - F_{e} \cdot x_{e}$$
⁽⁹¹⁾

Using the optimal EES operation from Section 5.2.1.1 and the corresponding Lagrange multipliers λ_d as price we can write:

$$\pi_{e} = v_{s} \int_{0}^{t_{s}} x_{e} dt + v_{p} \int_{t_{s}}^{t_{p}} x_{e} dt + v_{b} \int_{t_{p}}^{t_{b}} q_{e}(t) dt - \eta_{e} v_{b} \int_{t_{b}}^{t_{v}} q_{e-}(t) dt - F_{e} \cdot x_{e} =$$
(92)
$$x_{e} \cdot \left(v_{s}t_{s} + v_{p}(t_{p} - t_{s}) - F_{e}\right) + v_{b} \left(E_{e}^{[t_{p}, t_{b}]} - \eta_{e}E_{e-}^{[t_{b}, t_{v}]}\right)$$

Substituting the rightmost parenthesis with $\eta_e x_e (T - t_v) - x_e t_p$ from the annual storage balance (83) gives:

$$\pi_{e} = x_{e} \cdot \left(v_{s} t_{s} + v_{p} (t_{p} - t_{s}) + v_{b} (\eta_{e} (T - t_{v}) - t_{p}) - F_{e} \right)$$
(93)

Cost recovery $\pi_e = 0$ gives

$$v_{s}t_{s} + v_{p}(t_{p} - t_{s}) + v_{b}(\eta_{e}(T - t_{v}) - t_{p}) - F_{e} = 0$$

$$\Rightarrow t_{v} = T - \frac{1}{\eta_{e}v_{b}} \Big(F_{e} + v_{b}t_{p} - v_{s}t_{s} - v_{p}(t_{p} - t_{s}) \Big) = T - \frac{F_{b} - F_{e}}{\eta_{e}v_{b}}$$
(94)

which is equal to the system optimal condition found in (82) in the previous section.

Profit maximization of an individual EES owner who is sufficiently small to not change the duration t_v on its own gives the same result:

$$\frac{\partial \pi_e}{\partial x_e} = 0 \Longrightarrow v_s t_s + v_p (t_p - t_s) + v_b (\eta_e (T - t_v) - t_p) - F_e = 0$$
⁽⁹⁵⁾

5.2.1.3 Cost recovery of VRE

The general profit function for VRE is given in (56). Inserting the prices (Lagrange multipliers) derived from the optimal operation in Section 5.2.1.1, we get:

$$\pi_{v} = v_{s}AF_{v}^{[0,t_{s}]}x_{v}t_{s} + v_{p}AF_{v}^{[t_{s},t_{p}]}x_{v}(t_{p}-t_{s}) + v_{b}AF_{v}^{[t_{p},t_{b}]}x_{v}(t_{b}-t_{p}) + \eta_{e}v_{b}AF_{v}^{[t_{b},t_{v}]}x_{v}(t_{v}-t_{b}) - F_{v}x_{v}$$

$$(96)$$

Cost recovery, $\pi_{\nu} = 0$, gives the same result as system optimum (90):

$$F_{v} = p_{s}AF_{v}^{[0,t_{s}]}t_{s} + v_{p}AF_{v}^{[t_{s},t_{p}]}(t_{p} - t_{s}) + v_{b}AF_{v}^{[t_{p},t_{b}]}(t_{b} - t_{p}) + \eta_{e}v_{b}AF_{v}^{[t_{v},t_{e}]}(t_{v} - t_{b})$$

$$(97)$$

We see directly from (96) that this result is also obtained by maximizing the profit of an individual VRE owner in a perfect market: $\frac{d\pi_v}{dx_v} = 0$ gives the same result since π_v is a linear function of x_v .

5.2.2 VRE storage with limited energy capacity

As discussed in the introduction to this chapter, EES storage is only represented by its power capacity and efficiency since the duration curve model cannot handle energy capacity limitations of the storage explicitly. So far, we have assumed that it is possible to shift the stored energy between any time instants over the year, which obviously is a very optimistic assumption for both present and emerging storage technologies such as batteries, flow batteries, CAES and pumped hydro. These technologies are often reported with durations in the range of hours, both in real-world installations and future scenario studies (Pinnangudi, Kuykendal, and Bhadra 2017; Argyrou, Christodoulides, and Kalogirou 2018; Cole et al. 2016; McPherson, Johnson, and Strubegger 2018; Sepulveda et al. 2018; de Sisternes, Jenkins, and Botterud 2016). Although there are several promising alternatives for long-duration storage, these are so far limited by different factors, e.g. 1) Being at the R&D stage, such as Aqueous sulfur/sodium/air systems (Z. Li et al. 2017). 2) High round-trip losses such as hydrogen storage (Pellow et al. 2015; Korpås 2004). 3) Specific geographical locations such as seasonal-storage from reservoir hydro with pumping capability (Graabak et al. 2019).

In this section, we introduce an EES model that is limited to only discharge in the baseload period. The motivation behind this model is to represent short-term storage of e.g. PV energy from day to night, so-called "valley filling" strategy. The model is also used as a building-block of the general price arbitrage model (Section 5.2.3).



Figure 5. Net load duration curve for a system where stored VRE energy replaces baseload power only.

5.2.2.1 System optimality conditions

Based on the conservative assumption explained above, we can directly derive a simplified variant of the model presented in Section 5.2.1 where the storage only shifts energy between the surplus period

 $[t_b, T]$ to the base plant period $[t_p, t_b]$. In this case, neither the optimal duration nor the sizing of the peaker and base plant is affected by the inclusion of the storage. The optimality condition for the VRE plant is also unchanged. For EES, on the other hand, the periods when load shedding and peaker sets the marginal cost are no longer available for discharge, so the optimality condition similar to (85)-(87) becomes:

$$\eta_e v_b (T - t_v) = F_e \tag{98}$$

$$\implies t_v = T - \frac{F_e}{\eta_e v_b} \tag{99}$$

since the EES now discharges all its energy from surplus VRE to replace energy from the baseload plant in the period $[t_p, t_b]$.

5.2.2.2 Cost recovery

Cost recovery and profit maximization follows directly from the derivations in Section 5.2.1, by simply setting the EES profits to zero in the period $[0, t_p]$. The equivalents of (94) and (95) for this case both gives:

$$F_e = \eta_e v_b (T - t_v) \tag{100}$$

which is equal to the system optimum (98).

For the VRE-plant, the conditions for cost recovery and profit maximization is unchanged from Section 5.2.1, thus (97) applies also here.

5.2.3 Storage for general price arbitrage

In this section, we combine the limited VRE storage model from Section 5.2.2 with the storage model for conventional generators from Section 5.1. The rationale behind this combined model is to utilize ESS whenever it is profitable, whether that means storing thermal energy or renewable energy. In principle, a time-sequential optimization model based on the general formulation presented in Chapter 1 will seek cost-minimum without restricting the use of storage charging to any type of power plant. However, it is not possible to fully represent time-sequential storage optimization in a duration model approach due to the lack of chronology of the sorted net load. Based on the storage operations strategies presented previously, we propose in this section a combined storage model with the following operating strategy:

 $\langle t_b, T \rangle$: The EES stores surplus VRE energy and discharges it sometime during $[t_p, t_b]$. In Figure 6, this period is illustrated to be immediately before t_b , but it is not a necessary assumption as long as the storage is discharged sometime during the baseplant period. The operation strategy follows Section 5.2.2.

 $\langle t_e, t_b \rangle$: During this period, the baseload is the marginal generator. During a part of this period, the EES discharges surplus VRE energy, indicated by $E_{e,2}$ in Figure 6. In the rest of the period, the EES is available for charging baseload power to replace peaker and load shedding during $[0, t_p]$.

 $[0, t_e]$: The EES energy which was charged by the baseload is discharged in this period. Surplus energy from VRE is assumed not to be available for discharge in this period. This is a simplistic representation of the limited energy capacity (kWh capacity) of the storage, which also was used in Section 5.2.2.

For this EES representation to be practically applicable, the baseload period $[t_p, t_b]$ must be sufficiently long to have room for both discharging VRE energy from $[t_b, T]$ and to shift energy to the load shedding peaker periods. As an example, consider a system with peaker duration equal to 500 hours and with 2000 hours of surplus VRE energy. The period $[t_p, t_b]$ will then be 6260 hours, which is more than enough for the proposed storage operation. Nevertheless, the result should always be checked for this limitation.



Figure 6. Net Load duration curve including energy storage, which is charged by baseload $(E_{e-,1})$ and VRE $(E_{e-,2})$. The stored energy from the baseplant is discharged during load shedding and peaker operation $(E_{e,1})$. The stored energy from VRE replaces only baseload $(E_{e,2})$ due to the model assumption of limited storage capacity.

5.2.3.1 System optimality conditions

The EES operational model is a combination of two modes: 1) Store baseload energy $(E_{e-,1})$ to supply during load shedding and peaker operation $(E_{e,1})$. 2) Store surplus VRE energy $(E_{e-,2})$ to supply during baseload period $(E_{e,2})$. These operating and storage regimes do not interfere, and occurs at different periods, so we can treat them separately when deriving the optimal system operation.

 $[0, t_s], \langle t_s, t_p], \langle t_p, t_e]$: The optimal operation in the these intervals is identical to what is stated in Table 3, except that the thermal generators and EES must meet the net demand instead of demand since VRE is present.

 $(t_e, t_b]$: The baseload is the marginal generator, both when EES is charging from the baseload (See Table 3) and when it is discharging previously stored VRE energy (as explained in Section 5.2.1.1).

 $\langle t_b, t_v \rangle$, $\langle t_v, T \rangle$: The optimal operation is the same as in Section 5.2.1.1 for these intervals.

Based on the optimal operation in the different duration segments according to Figure 6, we can set up the system cost minimization problem in the same manner as previously:

$$\min_{i} C = \sum_{i} F_{i} x_{i} + v_{s} \int_{0}^{t_{s}} [q_{d}(t) - AF_{v}(t)x_{v} - \sum_{j} x_{j}] dt + v_{p} \cdot \left(\int_{0}^{t_{s}} x_{p} dt + \int_{t_{s}}^{t_{p}} [q_{d}(t) - AF_{v}(t)x_{v} - \sum_{k} x_{k}] dt\right) + v_{b} \cdot \left(\int_{0}^{t_{e}} x_{b} dt + \int_{t_{e}}^{t_{b}} [q_{d}(t) - AF_{v}(t)x_{v} + q_{e-}(t) - q_{e}(t)] dt\right) \forall i \in \{p, b, v, e\}, j \in \{p, b, e\}, k \in \{e, b\}$$
(101)

s.t.
$$\int_{t_e}^{t_b} q_{e^-}(t) dt = \eta_e^{-1} x_e t_p + \eta_e^{-1} \int_{t_p}^{t_e} q_e(t) dt$$
 (102)

$$\int_{t_e}^{t_b} q_e(t) dt = \eta_e \int_{t_b}^{t_v} q_e(t) dt + \eta_e x_e(T - t_v)$$
(103)

By setting $\int_{t_e}^{t_b} q_{e-}(t) dt = E_{e-}^{[t_e, t_b]}$, and applying storage conservation of the charged baseload energy, we can formulate (102):

$$E_{e^{-}}^{[t_e,t_b]} = \eta_e^{-1} x_e t_p + \eta_e^{-1} E_e^{[t_p,t_e]}$$

$$= \eta_e^{-1} x_e t_p + \eta_e^{-1} E_d^{[t_p,t_e]} - \eta_e^{-1} (x_b + A F_v^{[t_p,t_e]} x_v) \cdot (t_e - t_p)$$
(104)

Similarly, for the VRE storage constraint (103):

$$E_{e}^{[t_{e},t_{b}]} = \eta_{e} A F_{v}^{[t_{b},t_{v}]} x_{v} - \eta_{e} E_{d}^{[t_{b},t_{v}]} + \eta_{e} x_{e} (T - t_{v})$$
(105)

As previously, we write for convenience $E_d^{[0,t_s]} = \int_0^{t_s} q_d(t)dt$ and $AF_v^{[0,t_s]} = \int_0^{t_s} AF_v(t)dt$ etc. We insert the reformulated storage constraints (104) and (105) into the objective function (102) and rearrange with respect to the generator capacities x_i . The cost minimization problem can then be written:

$$\min_{i} C = \sum_{i} a_{i} x_{i} + b \ \forall i \in \{p, b, e, v\}$$

$$(106)$$

where

$$b = v_s E_d^{[0,t_s]} + v_p E_d^{[t_s,t_p]} + \eta_e^{-1} v_b E_d^{[t_p,t_e]} + v_b E_d^{[t_e,t_b]} + \eta_e v_b E_d^{[t_b,t_v]}$$
(107)

$$a_p = F_p - (v_s - v_p)t_s \tag{108}$$

$$a_b = F_b - v_s t_s - v_p (t_p - t_s) - v_b \eta_e^{-1} (t_e - t_p) + v_b t_e$$
(109)

$$a_{v} = F_{v} - v_{s}AF_{v}^{[0,t_{s}]}t_{s} - v_{p}AF_{v}^{[t_{s},t_{p}]}(t_{p} - t_{s}) - \eta_{e}^{-1}v_{b}AF_{v}^{[t_{p},t_{e}]}(t_{e} - t_{p}) -$$

$$(110)$$

$$u_{s}AF_{v}^{[t_{e},t_{b}]}(t_{e} - t_{e}) - u_{s}AF_{v}^{[t_{b},t_{v}]}(t_{e} - t_{e})$$

$$a_e = F_e - v_s t_s - v_p (t_p - t_s) + v_b \eta_e^{-1} t_p - v_b \eta_e (T - t_v)$$
(111)

Since a_i is the partial derivate of *C* with respect to x_i , the optimality condition for the investment in each technology $i \in \{p, b, v, e\}$ is found by setting $a_i = 0$ in (108)-(111). t_s is solved directly from (108) as before. We then have three equations left with four unknowns (t_p, t_b, t_e, t_v) . However, the durations are also constrained by the net demand curve $q_{nd}(t) = q_d(t) - AF_v(t) \cdot x_v$. The net demand is zero at $t = t_b$:

$$q_{nd}(t_b) = q_d(t_b) - AF_v(t_b) \cdot x_v = 0 \Rightarrow x_v = \frac{q_d(t_b)}{AF_v(t_b)}$$
(112)

Equation (112) expresses x_v as a function of t_b . Furthermore, the net demand is equal to $-x_e$ at $t = t_v$:

$$x_e = -q_{nd}(t_v) = -q_d(t_v) + AF_v(t_v) \cdot x_v$$
(113)

From the left part of Figure 6, we see that x_e is also bounded by:

$$x_{e} = q_{nd}(t_{p}) - q_{nd}(t_{e}) = q_{d}(t_{p}) - AF_{v}(t_{p}) \cdot x_{v} - q_{d}(t_{e}) + AF_{v}(t_{e}) \cdot x_{v}$$
(114)

By combining (112), (113) and (114) we get a non-linear equation $f(t_p, t_b, t_e, t_v) = 0$ which bounds the solutions to the net demand curve, Thus, there are in total four equations with four unknowns which determines the optimality conditions for EES, VRE and the baseplant:

$$0 = F_b - v_s t_s - v_p (t_p - t_s) - v_b \eta_e^{-1} (t_e - t_p) + v_b t_e$$
(115)

$$0 = F_{v} - v_{s}AF_{v}^{[0,t_{s}]}t_{s} - v_{p}AF_{v}^{[t_{s},t_{p}]}(t_{p} - t_{s}) - \eta_{e}^{-1}v_{b}AF_{v}^{[t_{p},t_{e}]}(t_{e} - t_{p}) -$$
(116)

$$0 = F_e - v_s t_s - v_p (t_p - t_s) + v_b \eta_e^{-1} t_p - v_b \eta_e (T - t_v) = 0$$
(117)

$$0 = q_d(t_v) + q_d(t_p) - q_d(t_e) - \frac{q_d(t_b)}{AF_v(t_b)} \cdot \left(AF_v(t_v) + AF_v(t_p) - AF_v(t_e)\right)$$
(118)

This set of equations can be solved with respect to the unknowns $t_i \forall i \in \{p, b, e, v\}$ by a suitable iterative procedure or non-linear equation solver. i.e. the bisection method. When the durations are found, it is straightforward to calculate the optimal capacities $x_i \forall i \in \{p, b, e, v\}$ from the net demand duration curve.

5.2.3.2 Cost recovery and profit maximization

According to the optimal operating strategy derived in the previous section, there are now six periods with different prices obtained from the Lagrange multiplier λ_d , displayed in Table 4.

Table 4. Price (λ_d) periods for the EES model with general price arbitrage.

M. Korpås, A. Botterud. Optimality Conditions and Cost Recovery in Electricity Markets with Variable Renewable Energy and Energy Storage, MIT CEEPR Working Paper 2020-005, March 2020.

Period	$[0, t_{s}]$	$\langle t_s, t_p]$	$\langle t_p, t_e]$	$\langle t_e, t_b]$	$\langle t_b, t_v]$	$\langle t_v, T]$
λ_d	v_s	v_p	$\eta_e^{-1} v_b$	v_b	$\eta_e v_b$	0

The cost recovery and profit maximization results for the all technologies can be summarized as follows:

Peaker: Operation is similar to the previous sections, so results from Section 3.2 also applies here.

<u>Baseload</u>: Operation is similar to the case with EES but no VRE (Section 5.1.2), except that there are some discharge of stored energy from VRE during $\langle t_e, t_b \rangle$. However, this does not change the profit of the baseload, because it is the marginal generator during $\langle t_e, t_b \rangle$. Therefore, the profit function is equal to the one found in Section 5.1.2, and the results applies here as well.

<u>VRE</u>: Profits are similar to Section 5.2.1.3, expect that there is an additional price segment $\langle t_p, t_e \rangle$ to account for:

$$\pi_{v} = \left(v_{s}AF_{v}^{[0,t_{s}]}t_{s} + v_{p}AF_{v}^{[t_{s},t_{p}]}(t_{p}-t_{s}) + \eta_{e}^{-1}v_{b}AF_{v}^{[t_{e},t_{b}]}(t_{b}-t_{e}) + v_{b}AF_{v}^{[t_{e},t_{b}]}(t_{b}-t_{e}) + \eta_{e}v_{b}AF_{v}^{[t_{b},t_{v}]}(t_{v}-t_{b}) - F_{v}\right) \cdot x_{v}$$

$$(119)$$

Cost recovery and profit maximization both yields:

$$v_{s}AF_{v}^{[0,t_{s}]}t_{s} + v_{p}AF_{v}^{[t_{s},t_{p}]}(t_{p}-t_{s}) + \eta_{e}^{-1}v_{b}AF_{v}^{[t_{p},t_{e}]}(t_{e}-t_{p}) + v_{b}AF_{v}^{[t_{e},t_{b}]}(t_{b}-t_{e})$$

$$+ \eta_{e}v_{b}AF_{v}^{[t_{b},t_{v}]}(t_{v}-t_{b}) - F_{v} = 0$$

$$(120)$$

which is the equal to the optimality condition for the VRE found in Section 5.2.3.1.

<u>EES</u>: The market operation is now a combination of the optimal strategies from Section 5.1.2 and Section 5.2.2.2 with profit function:

$$\pi_{e} = v_{s} \int_{0}^{t_{s}} x_{e} dt + v_{p} \int_{t_{s}}^{t_{p}} x_{e} dt - v_{b} \int_{t_{e}}^{t_{b}} q_{e-}(t) dt + v_{b} \int_{t_{e}}^{t_{b}} q_{e}(t) dt -$$
(121)
$$\eta_{e} v_{b} \int_{t_{b}}^{t_{v}} q_{e-}(t) dt - F_{e} \cdot x_{e}$$

Substituting $\int_{t_b}^{t_v} q_{e^-}(t) dt$ and $\int_{t_b}^{t_v} q_e(t) dt$ with the storage conservation equations (104) and (105) yields:

$$\pi_e = (v_s t_s + v_p (t_p - t_s) - v_b \eta_e^{-1} t_p + v_b \eta_e (T - t_v) - F_e) \cdot x_v$$
(122)

We see that both cost recovery ($\pi_v = 0$) and profit maximization ($\frac{d\pi_v}{dx_v} = 0$) lead to the system optimal condition from Section 5.2.3.1.

6 Numerical example

6.1 Case study description

To illustrate potential use of the proposed model we constructed a numerical example based on European energy and cost data from the EU 2050 Reference Scenario (European Commission 2016) and an associated technology data set from the EU-project ASSET (De Vita, Kielichowska, and Mandatowa 2018). The analyzed cases are listed in Table 5, where the case numbering follows the order of appearance in Chapters 3-5.

Case	Peaker	Base	VRE	EES	EES model	Chapter
1	х	Х			-	3
2	х	Х	Х		-	4
3	х	Х		х	EES for thermal plants	5.1
4	х	Х	х	х	Unlimited VRE storage	5.2.1
5	х	Х	Х	х	Limited VRE storage	5.2.2
6	х	Х	Х	х	General price arbitrage	5.2.3

Table 5. Case study description.

The peaker and baseload technologies are chosen to be OCGT and CCGT, respectively. Simple OCGT plant data are not included in the EU Reference data, so we set the capital cost was set to 50 % of the CCGT capital cost, based on data from the U.K. Dept. of Energy and Climate Change, 2013 (Parsons Brinckerhoff 2013). Offshore wind is implemented as the VRE technology, while stationary Li-Ion Battery Energy Storage System (BESS) and Pumped Hydro Energy Storage (PHES) are used as examples of EES. Costs of Li-ion BESS are based on the low cost assumptions from (Cole et al. 2016).

We use the Norwegian reservoir hydropower system as basis for setting up the PHES model parameters. Several papers have described the possibilities for extending parts of the Norwegian hydro system by installing new reversible pump systems between large existing reservoirs (Harby et al. 2013; Korpås, Wolfgang, and Aam 2015; Askeland, Jaehnert, and Korpås 2019; Graabak et al. 2019). Previous studies have revealed that there is a large potential for utilizing the existing hydro reservoirs with additional pumping capability and increase the capacity of offshore HVDC connections to balance North Sea offshore wind and to connect to the continental European power system. In our case study, we analyze the investment of pumping capacity to the existing seasonal hydro storage system in Norway based on cost and technology data from (Korpås, Wolfgang, and Aam 2015). This means that this EES option has all its costs associated with power capacity, assuming seasonal storage by existing reservoirs.

We use load time series from the ENTSO-E 2040 GCA dataset for the following European countries: AT, BE, BG, CH, CZ, DE, DK, EE, ES, FI, FR, EL, HR, HU, IE, IT, LT, LU, LV, NL, NO, PL, PT, RO, SI, SK, SE, UK. Offshore Wind production data is provided by the JRC EMHIRES data set (European Commission 2019) for the countries above with shoreline. We use generation capacities from the ENTSO-E 2040 Global Climate Action Scenario (ENTSO-E 2018) for the same countries. Aggregated offshore wind time-series are given in p.u. of the total installed capacities. All time-series represent historical weather and load conditions for year 2007. We set the scarcity price in the market to 3000 €/MWh, which is equal to the current maximum price in the EPEX spot market (http://www.epexspot.com).

The cost and technology data are summarized in Table 6. Variable cost of thermal generation is calculated from

$$v_i = \eta_i^{-1} \left(p_{fuel,i} + p_{CO2} EF_{fuel,i} \right) + OM_{var,i}$$
(123)

The resulting variable costs are 155.0 \notin /MWh and 103.2 \notin /MWh for the peaker and the baseplant, respectively. Annual fixed costs are calculated from $F_i = SCC_i \cdot \alpha_i + OM_{fix,i}$ where α_i is the annuity factor, $\alpha_i = r \cdot (1 - (1 + r)^{-L_i})^{-1}$. Since energy capacity restrictions of the EES are not handled explicitly in the duration curve model, it is necessary to assume a certain energy capacity to calculate the total fixed cost of EES per kW. This is done by specifying the storage duration *d* in the BEES cost function provided in Table 6. For the PHES model, this is not necessary since the energy storage capacity is assumed to be the existing one, and all investment costs are associated with expanding the existing hydro plant with reversible pumps and necessary extensions of grid infrastructure.

Table 6. Cost data and power system data for the numerical example. e1 is BESS, e2 is PHES. The total cost of e1 depends on its assumed duration d.

Parameter	Value	Unit	Parameter	Value	Unit
$p_{fuel,i}$, $i \in [p, b]$	[48.5 48.5]	€/MWh _{fuel}	v_s	3000	€/MWh
$OM_{var,i}$, $i \in [p, b, v, e]$	[1.73 1.73 0 0]	€/MWh	p_{CO2}	63	€/tonco2
$OM_{fix,i}$, $i \in [p, b, v, e]$	[15 15 40.7 0]	€/MW/yr	\overline{P}_d	100	MW
$EF_{fuel,i}$, $i \in [p, b]$	[0.18 0.18]	ton_{CO2}/MWh_{fuel}	r	8.5	%
$L_i, i \in [p, b, v, e]$	[30 30 25 15]	yr			
$\eta_i, i \in [p, b, e]$	[39 59 81]	%			
$SCC_i, i \in [p, b, v]$	[320 640 1891]	€/kW			
SCC _{e1}	265+d·65	€/kW			
SCC _{e2}	2500	€/kW			

The analytical solutions derived in chapters 3-5 showed that the market solution gives the optimal solution for all the cases listed in Table 5. Thus, the Weighted Average Price of Electricity (WAPE) for the costumers in the market will be equal to the Annual Cost of Electricity (ACE):

$$ACE_{op} = \frac{v_s \int_0^{t_s} q_s(t)dt + v_p \int_0^{t_p} q_p(t)dt + v_b \int_0^{t_b} q_b(t)dt}{\int_0^T q_d(t)dt} = \frac{v_s E_s + v_p E_p + v_b E_b}{E_d}$$
(124)

$$ACE_{fix} = \frac{\sum_{i \in \{p, b, v, e\}} F_i x_i}{E_d}$$
(125)

$$ACE = ACE_{op} + ACE_{fix}$$
(126)

The general equation for WAPE used here is

$$WAPE = \frac{\int_0^T p(t)q_d(t)dt}{E_d}$$
(127)

where the price p(t) is given by the marginal generator/consumer of electricity at time instant (t). For the most general case with EES, Case 6, WAPE becomes:

$$WAPE = \frac{v_{s} \int_{0}^{t_{s}} q_{d}(t)dt + v_{p} \int_{t_{s}}^{t_{p}} q_{d}(t)dt + \frac{v_{b}}{\eta_{e}} \int_{t_{p}}^{t_{e}} q_{d}(t)dt + v_{b} \int_{t_{e}}^{t_{b}} q_{d}(t)dt + \eta_{e}v_{b} \int_{t_{b}}^{t_{v}} q_{d}(t)dt}{E_{d}}$$
(128)

In all cases WAPE = ACE so we only report ACE in the results for convenience.

Finally, we use the Levelized Cost of Electricity (LCOE) for VRE to analyze the economic impact of curtailment:

$$LCOE_{vpot} = \frac{(F_v + OM_{fix,v})x_v}{(T_a - (t))t_t} = \frac{F_v + OM_{fix,v}}{(T_a - (t))t_t}$$
(129)

$$LCOE_{v} = \frac{(F_{v} + OM_{fix,v})x_{v}}{\int_{0}^{t_{b}} q_{v}(t)dt} = \frac{F_{v} + OM_{fix,v}}{\int_{0}^{t_{b}} AF_{v}(t)dt}$$
(130)

where $q_{mot}(t)$ is the potential VRE output in time instant t in the absence of curtailment.

6.2 Results without EES (Case 1 and 2)

The main results for the two cases without EES are given in Table 7. From Case 2 we see that VRE is competitive in the market, with an optimal capacity of 94.4 % of the peak demand, resulting in a VRE share of 62.1 %. There is surplus VRE generation, and the price becomes zero in 31 % of the year. VRE pushes some baseplant capacity out of the market and gives room for some more peaker capacity. This shift in capacities is visualized in Figure 7 where the duration curves for Case 1 and Case 2 are plotted together. With the assumed cost estimates for 2050, unsubsidized VRE reduces the average cost of energy for the customers by almost 30 %. Note that the number of hours with load shedding t_s is the same in two cases due to the optimality condition for the peaking plant, which does not change with the introduction of VRE. However, there is an increase in the energy not supplied (ENS) and maximum load shedding in Case 2. This effect of VRE is unavoidable given that 1) there are no entry of new technologies, 2) demand is inelastic, 3) all generators must recover their costs in the energy-only market. Changes in reserve requirements and/or capacity payments due to VRE integration may lead to a different ENS result, but such analyses are left for future works using models with more detailed representation of system operation.

Table 7. Results for Case 1 (Only thermal) and Case 2 (Thermal+VRE). In Case 2, the energy share of wind is 62.1 % of the annual demand. Wind curtailment is 9.5 % of the annual available wind energy.

	t _s	t_p	t_b	x_p	x_b	x_v	ACE	$max(q_s)$	ENS
Case	[h]	[ĥ]	[h]	[MW]	[MW]	[MW]	[€/MWh]	[MW]	[%]
1	15.7	572.5	8760	8.2	89.1	-	115.0	2.4	0.003
2	15.7	572.5	6029	15.0	66.7	94.3	81.6	11.4	0.008



Figure 7. Duration curves for Case 1 and Case 2. The duration of load shedding is too short to be clearly visible. Price segments are placed beneath the x-axis.

The results presented above are obtained for a specific 2050 scenario for offshore wind in Europe, where future investment costs are obviously very uncertain. It is therefore important to study how the optimal capacities and durations is influenced by changes in VRE investment cost. In Figure 8, we have plotted the optimal installed VRE capacity as a function of its investment cost. The shape of the curve is interesting; an almost exponential increase as the cost decreases from 5000 to 4500 €/kW, followed by a long segment with linear increase. Finally, at very low investment costs, the VRE capacity increases at a higher pace again. Detailed analysis of this effect is provided in Appendix A.



Figure 8. Optimal installed capacity of VRE as a function of its specific investment cost. The result of the selected EU cost scenario for 2050 is shown with dotted lines.

From Figure 9 (upper), we see that the VRE energy share follows in general the same pattern as its capacity when the investment cost comes down. However, at very low investment costs, the energy share increases more moderately due to the unavoidable curtailment of excess wind generation. In the same figure (lower), we have also plotted the energy share as function of installed capacity, where we recognize the expected saturation of utilized VRE energy as the capacity increases beyond the minimum demand of 42 MW.

Figure 10 shows ACE and LCOE as functions of VRE investment costs. The LCOE of the baseload plant first start to increase noticeably when its duration becomes less than 8760 hours, which occurs for VRE costs lower than 4500 €/kW. At very low VRE investment cost, the ACE starts to tip downwards.

This might seem as a surprising result, as since the VRE curtailment increases at the same time. However, this effect can be explained by recalling the non-linear increase in optimal VRE capacity in the same cost region (see rightmost part of Figure 8 and detailed discussion in Appendix A). If we instead plot the ACE as a function of optimal VRE capacity, we see that the ACE reduction slows down at the higher capacities due to the curtailment of surplus VRE generation (Figure 11).



Figure 9. VRE share of supply (blue) and VRE curtailment (red) as a function of VRE investment cost (upper) and installed capacity (lower). The optimal result of the selected EU 2050 cost scenario is shown with dotted lines.



Figure 10. Average Cost of Electricity (ACE) and Levelized Cost of Electricity (LCOE) as a function of VRE investment cost. The result of the selected EU 2050 cost scenario is shown with dotted lines.



Figure 11. Average Cost of Electricity as a function of optimal VRE capacity.

6.3 **Results with EES**

6.3.1 Case 3: Peaker+Baseload+EES

Here, EES is only used for shifting energy from the base plant period to the peaker and load shedding periods. The highest possible fixed cost F'_e for EES to be feasible can be found from (72). With the parameters in Table 6, $F'_e = 60.7 \notin kW/yr$, corresponding to $SCC'_e = 504 \notin kW$. Using the BESS cost formula from Table 6 ($SCC_e = 265 + d \cdot 64$) the maximum economically feasible storage duration becomes d = 3.74 h.

We found in the previous section that $ACE = 115 \notin MWh$ with only peaker and baseplant in the system. If we add BESS with very optimistic cost estimate $SCC_e = 100 \notin kW$, the ACE is only reduced by 1 %. This is because the EES does not contribute to more than a modest amount of load shifting between thermal generators which highly limits the cost reduction potential for the system. Therefore, the cases with VRE is more interesting, since EES can facilitate a higher VRE share and thus a significant change in system characteristics.

6.3.2 Case 4: Peaker+Baseload+VRE+EES "Unlimited VRE storage"

In this case, the EES is only charged by surplus VRE, and it can be discharged at any other period of the year. The optimality condition for the EES (87) determines the duration of the year t_v when the available VRE is fully utilized. Equation (87) states that that t_v is negatively correlated with the EES investment cost. The criterion for EES to be profitable is when t_v is equal to the baseload duration for the case without EES: $F'_e = F_b + \eta_e v_b (T - t_v)$. From this relation, we can find the maximum acceptable investment cost $SCC'_e = F'_e/\alpha_e$. Inserting the case study parameters from Table 6, we get $SCC'_e = 2514 \notin /kW$. This is very close to the cost estimate of PHES of 2500 \notin /kW , indicating that a PHES investment would only have been marginally profitable in this case. If we use the BESS cost formula from Table 6 instead, the maximum economically feasible storage duration becomes d = 35, which may be realistic for flow batteries in the future. For comparison, the generation expansion studies performed in (Cole et al. 2016; de Sisternes, Jenkins, and Botterud 2016; Cebulla, Naegler, and Pohl 2017) reports Li-Ion BESS storage durations in the range of 2-8 hours.



Figure 12. Upper figure: Durations t_v and t_b as functions of EES investment cost SCC_e . Lower figure: Capacities x_v , x_e and x_b as functions of SCC_e . The dotted line is the threshold cost of EES for being profitable.

In the upper part Figure 12, we plot t_v and t_b as functions of EES investment cost, where the threshold cost SCC'_e also is marked. This is the point where t_v and t_b diverge, and we observe that the baseload duration t_b is much more sensitive to the EES cost than t_v . The resulting installed capacities are shown in the lower part of Figure 12. It is evident that EES triggers significant amounts of additional VRE capacity. This creates a double negative effect on the baseload capacity as the EES costs declines: Both the EES itself and the additional VRE capacity replace the baseload plant. The peaker capacity (not shown in the figure) is not noticeably affected (i.e. x_p is around 15 MW for all EES costs, as in the VRE-only case). This is because it is still needed together with the EES in the peak hours, being less capital intensive than the baseplant.

An interesting observation from Figure 13 (upper) is that the impact of EES investment cost on the ACE is almost negligible, as opposed to the impact of VRE cost, which has causes a much larger reduction, as shown in Figure 10. Hence, our results indicate that cost reductions in clean generation technologies are more important than cost reductions in balancing technologies. On the other hand, EES expansion makes a significant impact on the VRE share, and indirectly on emissions, since it both replaces thermal generation for balancing and causes an increase in system optimal VRE investment. This is illustrated in the lower part of Figure 13. From this figure, we also see that EES lead to less relative curtailments of VRE. Moreover, the total amounts of curtailed VRE (in MWh/year) also decreases, although the EES triggers more VRE capacity in the system.



Figure 13. Upper: Average Cost of Electricity. Lower: VRE share and VRE curtailment as a function of EES investment cost SCC_e .

6.3.3 Case 5: Peaker+Baseload+VRE+EES "Limited VRE storage"

We only briefly report results from this special case, where the EES is assumed to be limited to discharge during the baseplant period. The rationale behind this assumption is that it is a "worst-case", contrasting the much more optimistic assumption behind Case 4 above. The threshold EES investment cost is obtained by setting $t_v = t_b$ in the optimality condition (99). Using the input data from Table 6, we get:

$$\begin{aligned} F'_e &= \eta_e v_b (T - t_b) = 0.81 \cdot 103.2 \cdot (8760 - 6029) = 22.821 \frac{\notin}{\text{MW yr}} \\ SCC'_e &= \frac{F'_e}{\alpha_e} = 1000 \cdot \frac{22.821}{0.1204} = 1.895 \frac{\notin}{\text{kW}} \\ d' &= \frac{(SCC'_e - 265)}{65} = 25.5 \text{ h} \end{aligned}$$

Not surprisingly, the threshold $\cot SCC'_e$ is lower than in Case 4, and considerably lower than the PHES investment $\cot (2500 \ e/kW)$, since the EES creates less value for the system here by only replacing baseload capacity. Equivalently, the maximum affordable energy capacity of BESS, measured in storage duration, is 25.5 h, compared to 33 h in Case 4.

6.3.4 Case 6: Peaker+Baseload+VRE+EES "General price arbitrage"

The last EES model that has been developed here combines the "limited VRE storage" of Case 5 with Case 3, where some baseplant energy is charged by the battery to reduce load shedding and peaker generation. It is therefore a model that places in between Case 4 and Case 5 when it comes to how EES contributes to the system. The threshold EES cost cannot be calculated directly from the optimality conditions, since they form a non-linear set of equations with four unknowns $f(t_p, t_b, t_e, t_v) = 0$. Setting $t_v = t_b = 6029$ at the threshold of the EES investment, we get from (111):

$$\frac{\partial C}{\partial x_e} = a_e = 0 \Rightarrow F'_e = v_s t_s + v_p (t_p - t_s) - v_b \eta_e^{-1} t_p + v_b \eta_e (T - t_b) = 26\ 312\ \frac{\epsilon}{\text{MW yr}}$$
$$SCC'_e = 2\ 185\frac{\epsilon}{\text{kW}} , \quad d' = 30\ \text{h}$$

As expected, the threshold cost is between the corresponding values from Case 4 (most optimistic case) and Case 5 (most pessimistic case).

6.4 Comparison with Linear Programming formulation

To assess the proposed Duration Curve (DUR) model, we have also modelled the system as a timesequential Linear Programming (LP) problem based on the basic formulation (1)–(8) for comparison. Note that the LP model handles energy storage constraints explicitly by equation (8). We ran the model without activating this constraint, in order to represent the same system as in the DUR model for Cases 1-3. Moreover, to model the case of storing only surplus VRE (Case 4 from Section 5.2.1) in the LP model, we added the following charging constraint to the LP model:

$$q_{e-}(t) \le q_{v}(t) - q_{d}(t) \quad \forall t \in [1, T]$$
(131)

This constraint forces the charging power to be equal to or less than the surplus VRE power. The LP optimization model is implemented in Julia v. 0.64, using the JuMP and Clp libraries (https://julialang.org/). We compared the results of the DUR model with the LP model for Cases 1-4. Cases 5 and 6 are not directly comparable with the LP model due to the modelling of storage constrains, and therefore needs more extensive analyses which is a topic for future works.

For the EES cases, the DUR model gives a threshold investment cost of $504 \notin kW$ in Case 2 and 2514 $\notin kW$ in Case 4. We have therefore chosen different EES costs for Case 2 (425 $\notin kW$) and Case 4 (2377 \kW) to trigger some investments in EES in the respective cases. The EES cost of Case 2 is equivalent to a BESS with 2.5 h storage duration, while Case 4 is equivalent to a PHES plant with 5 % lower investment cost than the original data from Section 6.1. Table 8 presents the main results for both models. As expected, the results from LP and the duration curve model are identical in all four cases since they basically solve the same optimization problem for Cases 1 to 4. There are some negligible numerical discrepancies due LP discretization of the year into 8760 discrete time steps.

Case]	1		2		3	4	1
Model	DUR	LP	DUR	LP	DUR	LP	DUR	LP
Obj. func. (C) [k€/yr]	69.9	69.9	69.8	69.8	49.6	49.6	49.5	49.5
<i>t_s</i> [h]	15.7	16.0	15.7	16.0	15.7	16.0	15.7	16.0
t_p [h]	572.5	573.0	572.5	573.0	230.2	231.0	572.5	573.0
t_e [h]	-	-	-	-	966	966	-	-
t_b [h]	8760	8760	6029	6028	8760	8760	5288	5289
t_{v} [h]	-	-	-	-	-	-	6226	6226
x_p [MW]	8.5	8.5	15.0	15.0	5.4	5.4	15.2	15.2
x_{b} [MW]	89.1	89.1	66.7	66.7	85.9	85.9	52.5	52.5
x_{v} [MW]	-	-	94.3	94.3	-	-	107.9	107.9
x_e [MW]	-	-	-	-	6.2	6.2	13.2	13.2
ACE [€/MWh]	115.0	115.0	81.6	81.6	114.9	114.9	81.4	81.4
$\max(q_s)$ [MW]	2.4	2.4	11.4	11.4	2.4	2.4	11.9	11.9
ENS [%]	0.003	0.003	0.008	0.008	0.003	0.003	0.009	0.009
VRE share [%]	-	-	62.1	62.1	-	-	72.5	72.5
VRE curtail. [%]	-	-	9.5	9.5	-	-	7.7	7.7

Table 8. Results from the duration curve model and LP model for Cases 1-4.

Figure 14 shows the duration curves which have been extracted from the LP optimization results of Case 4. It corresponds to the generic duration curve in Figure 4. From this figure, it is possible to extract the optimal durations that are derived directly from the analytic model and corresponding capacities, as summarized in Table 8 for both models.



Figure 14. Duration curves extracted from the LP model for Case 4. Optimal capacities, prices, and durations are displayed in the figure. The VRE capacity is not visible in the figure, since it is indirectly given through the net load. Moreover, the duration of load shedding is too short (16 hours) to be visible.

7 Conclusions and further work

We presented an analytical approach to analyze system optimality conditions and cost recovery in systems with VRE and EES. We proposed a duration curve methodology based on existing theoretical models for systems with only thermal generation. We used the methodology to show that in all the modelled cases, all power plants recover their costs in system optimum. This result also applies for cases where surplus VRE gives periods with zero prices, and cases where EES sets the price either based on the marginal cost of charging or the marginal value of discharging, depending on the instantaneous power balance. Our analytical results show how VRE changes the capacity of thermal generation in equilibrium, causing less baseload capacity but more peaker capacity. The impacts of EES on optimal thermal and VRE capacity depends on how EES is operated and what limitations that are assumed on the energy storage capacity.

When EES is used for charging excess VRE that otherwise would have been curtailed, it triggers more VRE capacity in equilibrium, since the EES creates a new price segment based on the marginal value of storage, where the VRE gains additional profits. This result has several implications for the market equilibrium: 1) EES pushes more thermal capacity out of the market, both because of its balancing ability and because it triggers more investments in VRE, 2) EES leads to lower total amounts of curtailed VRE in equilibrium, although it triggers more VRE investments, 3) Numerical analyses indicates that the main benefit of EES is to increase the VRE share in the system and consequently further reduce emissions caused by thermal generation. The emission benefit is much more evident from the results than the impacts on electricity cost, even with high carbon prices in line with low-carbon scenarios for Europe stadium 2050. How a marginal economic benefit will impact the willingness to invest in merchant EES in electricity markets is a topic for further analyses.

Since the approach is based on the duration curve of the net demand, the model represents the energy limit of ESS is in an approximate way. We presented different EES representations which are suited for duration curve modelling, and the models were tested against LP optimization which gave identical results. The analytical models are well suited for expansion in several ways. First and foremost, more research is needed to explore improved ways of modelling the energy constraints of EES and how they impact market prices and optimal generation capacities. Another important simplification is the

assumption of deterministic input data for demand and VRE, and optimal EES operation assuming perfect foresight of the whole year. Hence, adding uncertainty to the formulation is another direction for future work. The models could also be enhanced to include operating reserve requirements.

Finally, we believe that customer participation in the market will be increasingly important as more of the generation is based on VRE technologies with zero marginal costs. Hence, crucial work lies ahead to include demand-side flexibility in analytical modelling frameworks as the one presented here, to gain further insights in how prices and optimal generation mix will evolve in markets based on marginal cost pricing in the future.

Acknowledgements

We would like to thank Dept. of Electric Power Engineering (NTNU), Faculty of Information Technology and Electrical Engineering (NTNU) and Laboratory of Information and Decision Systems (MIT) for supporting the collaboration that resulted in this research work. We would also like to thank our colleagues at NTNU and MIT for fruitful discussions that helped to improve this paper.

References

- Akhil, Abbas A, Georgianne Huff, Aileen B Currier, Benjamin C Kaun, Dan M Rastler, Stella Bingqing Chen, Andrew L Cotter, Dale T Bradshaw, and William D Gauntlett. 2015.
 "DOE/EPRI Electricity Storage Handbook in Collaboration with NRECA." www.sandia.gov/ess.
- Aneke, Mathew, and Meihong Wang. 2016. "Energy Storage Technologies and Real Life Applications – A State of the Art Review." *Applied Energy* 179 (October): 350–77. https://doi.org/10.1016/j.apenergy.2016.06.097.
- Arbabzadeh, Maryam, Ramteen Sioshansi, Jeremiah X. Johnson, and Gregory A. Keoleian. 2019. "The Role of Energy Storage in Deep Decarbonization of Electricity Production." *Nature Communications* 10 (1): 3413. https://doi.org/10.1038/s41467-019-11161-5.
- Argyrou, Maria C., Paul Christodoulides, and Soteris A. Kalogirou. 2018. "Energy Storage for Electricity Generation and Related Processes: Technologies Appraisal and Grid Scale Applications." *Renewable and Sustainable Energy Reviews* 94 (October): 804–21. https://doi.org/10.1016/j.rser.2018.06.044.
- Askeland, Magnus, Stefan Jaehnert, and Magnus Korpås. 2019. "Equilibrium Assessment of Storage Technologies in a Power Market with Capacity Remuneration." *Sustainable Energy Technologies and Assessments* 31 (February): 228–35. https://doi.org/10.1016/j.seta.2018.12.012.
- Bathurst, Graeme N., and Goran Strbac. 2003. "Value of Combining Energy Storage and Wind in Short-Term Energy and Balancing Markets." *Electric Power Systems Research* 67 (1): 1–8. https://doi.org/10.1016/S0378-7796(03)00050-6.
- Bothwell, Cynthia, and Benjamin F. Hobbs. 2017. "Crediting Wind and Solar Renewables in Electricity Capacity Markets: The Effects of Alternative Definitions upon Market Efficiency." *Energy Journal* 38: 173–88. https://doi.org/10.5547/01956574.38.SI1.cbot.
- Botterud, Audun, and Hans Auer. 2019. "Resource Adequacy with Increasing Shares of Wind and Solar Power: A Comparison of European and U.S. Electricity Market Designs." *Economics of Energy & Environmental Policy*, no. In Press.
- Botterud, Audun, Todd Levin, and Vladimir Koritarov. 2014. "Pumped Storage Hydropower: Benefits for Grid Reliability and Integration of Variable Renewable Energy." *Argonne National Laboratory* ANL/DIS-14. www.anl.gov.

- California Energy Commission. 2018. "Deep Decarbonization in a High Renewables Future" CEC-500-20 (June). https://www.ethree.com/wpcontent/uploads/2018/06/Deep_Decarbonization_in_a_High_Renewables_Future_CEC-500-2018-012-1.pdf.
- Cebulla, F., T. Naegler, and M. Pohl. 2017. "Electrical Energy Storage in Highly Renewable European Energy Systems: Capacity Requirements, Spatial Distribution, and Storage Dispatch." *Journal of Energy Storage* 14 (December): 211–23. https://doi.org/10.1016/j.est.2017.10.004.
- Cole, Wesley J., Cara Marcy, Venkat K. Krishnan, and Robert Margolis. 2016. "Utility-Scale Lithium-Ion Storage Cost Projections for Use in Capacity Expansion Models." In NAPS 2016 -48th North American Power Symposium, Proceedings. Institute of Electrical and Electronics Engineers Inc. https://doi.org/10.1109/NAPS.2016.7747866.
- Dell, Ronald M., and David A.J. Rand. 2001. "Energy Storage A Key Technology for Global Energy Sustainability." *Journal of Power Sources* 100 (1–2): 2–17. https://doi.org/10.1016/S0378-7753(01)00894-1.
- Denholm, Paul, Erik Ela, Brendan Kirby, and Michael Milligan. 2010. "The Role of Energy Storage with Renewable Electricity Generation." *National Renewable Energy Laboratory*, no. NREL/TP-6A2-47-187. http://www.osti.gov/bridge.
- Díaz-González, Francisco, Andreas Sumper, Oriol Gomis-Bellmunt, and Roberto Villafáfila-Robles. 2012. "A Review of Energy Storage Technologies for Wind Power Applications." *Renewable and Sustainable Energy Reviews* 16 (4): 2154–71. https://doi.org/10.1016/j.rser.2012.01.029.
- ENTSO-E. 2018. "Maps and Data." Ten Year Network Development Plan TYNDP 2018. 2018. https://tyndp.entsoe.eu/maps-data/.
- European Commission. 2016. "EU Reference Scenario 2016. Energy, Transport and GHG Emissions Trends to 2050." https://doi.org/10.2833/9127.
 - ——. 2019. "EMHIRES Datasets | SETIS European Commission." SETIS Strategic Energy Technologies Information System. 2019. https://setis.ec.europa.eu/EMHIRES-datasets.
- Førsund, Finn R. 2015. Hydropower Economics. Springer.
- Fosso, Olav B., Anders Gjelsvik, Arne Haugstad, Birger Mo, and Ivar Wangensteen. 1999. "Generation Scheduling in a Deregulated System. the Norwegian Case." *IEEE Transactions on Power Systems* 14 (1): 75–80. https://doi.org/10.1109/59.744487.
- Gabriel, Steven A., Antonio J. Conejo, J. David Fuller, Benjamin F. Hobbs, and Carlos Ruiz. 2013. Complementarity Modeling in Energy Markets. International Series in Operations Research & Management Science. Vol. 180. Springer. https://doi.org/10.1007/978-1-4419-6123-5.
- Graabak, Ingeborg, Magnus Korpås, Stefan Jaehnert, and Michael Belsnes. 2019. "Balancing Future Variable Wind and Solar Power Production in Central-West Europe with Norwegian Hydropower." *Energy*, February, 870–82. https://doi.org/10.1016/j.energy.2018.11.068.
- Green, Richard. 2000. "Competition in Generation: The Economic Foundations." *Proceedings of the IEEE* 88 (2): 128–39. https://doi.org/10.1109/5.823994.
- Green, Richard, and Iain Staffell. 2015. "The Long-Run Impact of Energy Storage on Electricity Prices and Generating Capacity." In 38th IAEE Int. Conf. - Energy Security, Technology and Sustainability Challenges Across the Globe. Antalya, Turkey: IAEE. https://www.iaee.org/en/publications/newsletterdl.aspx?id=286.
- Harby, Atle, Julian Sauterleute, Magnus Korpås, Ånund Killingtveit, Eivind Solvang, and Torbjørn Nielsen. 2013. "Pumped Storage Hydropower." In *Transition to Renewable Energy Systems*, 597–618. Wiley Blackwell. https://doi.org/10.1002/9783527673872.ch29.

- Hasan, Ebrahim, Francisco D. Galiana, and Antonio J. Conejo. 2008. "Electricity Markets Cleared by Merit Order - Part I: Finding the Market Outcomes Supported by Pure Strategy Nash Equilibria." *IEEE Transactions on Power Systems* 23 (2): 361–71. https://doi.org/10.1109/TPWRS.2008.919238.
- Helm, Carsten, and Mathias Mier. 2016. "Efficient Diffusion of Renewable Energies: A Roller-Coaster Ride." Oldenburg Discussion Papers in Economics V-389–16. https://econpapers.repec.org/paper/olddpaper/389.htm.

——. 2018. "Subsidising Renewables but Taxing Storage? Second-Best Policies with Imperfect Carbon Pricing." Oldenburg Discussion Papers in Economics V-413–18. https://econpapers.repec.org/paper/olddpaper/413.htm.

- Hiroux, Celine, and Marcelo Saguan. 2010. "Large-Scale Wind Power in European Electricity Markets: Time for Revisiting Support Schemes and Market Designs?" *Energy Policy* 38 (7): 3135–45. https://doi.org/10.1016/J.ENPOL.2009.07.030.
- Hirth, Lion. 2013. "The Market Value of Variable Renewables. The Effect of Solar Wind Power Variability on Their Relative Price." *Energy Economics* 38 (July): 218–36. https://doi.org/10.1016/j.eneco.2013.02.004.
- Joskow, Paul L. 1976. "Contributions to the Theory of Marginal Cost Pricing." *Bell J Econ* 7 (1): 197–206. https://doi.org/10.2307/3003196.
- ———. 2019. "Challenges for Wholesale Electricity Markets with Intermittent Renewable Generation at Scale: The US Experience." Oxford Review of Economic Policy 35 (2): 291–331. https://doi.org/10.1093/oxrep/grz001.
- Kalkuhl, Matthias, Ottmar Edenhofer, and Kai Lessmann. 2013. "Renewable Energy Subsidies: Second-Best Policy or Fatal Aberration for Mitigation?" *Resource and Energy Economics* 35 (3): 217–34. https://doi.org/10.1016/j.reseneeco.2013.01.002.
- Korpås, Magnus. 2004. "Distributed Energy Systems with Wind Power and Energy Storage." *PhD Thesis*. Norwegian University of Science and Technology. http://hdl.handle.net/11250/249683.
- Korpås, Magnus, Arne T. Holen, and Ragne Hildrum. 2003. "Operation and Sizing of Energy Storage for Wind Power Plants in a Market System." In *International Journal of Electrical Power and Energy System*, 25:599–606. https://doi.org/10.1016/S0142-0615(03)00016-4.
- Korpås, Magnus, Ove Wolfgang, and Sverre Aam. 2015. "Norwegian Pumped Hydro for Providing Peaking Power in a Low-Carbon European Power Market - Cost Comparison against OCGT and CCGT." In *International Conference on the European Energy Market, EEM.* https://doi.org/10.1109/EEM.2015.7216688.
- Levin, Todd, and Audun Botterud. 2015. "Electricity Market Design for Generator Revenue Sufficiency with Increased Variable Generation." *Energy Policy* 87 (December): 392–406. https://doi.org/10.1016/j.enpol.2015.09.012.
- Li, Nan, Canan Uckun, Emil M. Constantinescu, John R. Birge, Kory W. Hedman, and Audun Botterud. 2016. "Flexible Operation of Batteries in Power System Scheduling with Renewable Energy." *IEEE Transactions on Sustainable Energy* 7 (2): 685–96. https://doi.org/10.1109/TSTE.2015.2497470.
- Li, Zheng, Menghsuan Sam Pan, Liang Su, Ping Chun Tsai, Andres F. Badel, Joseph M. Valle, Stephanie L. Eiler, Kai Xiang, Fikile R. Brushett, and Yet Ming Chiang. 2017. "Air-Breathing Aqueous Sulfur Flow Battery for Ultralow-Cost Long-Duration Electrical Storage." *Joule* 1 (2): 306–27. https://doi.org/10.1016/j.joule.2017.08.007.
- Mcdaniel, G. H., and Anthony F. Gabrielle. 1966. "Dispatching Pumped Storage Hydro." *IEEE Transactions on Power Apparatus and Systems* PAS-85 (5): 465–71.

https://doi.org/10.1109/TPAS.1966.291683.

- McPherson, Madeleine, Nils Johnson, and Manfred Strubegger. 2018. "The Role of Electricity Storage and Hydrogen Technologies in Enabling Global Low-Carbon Energy Transitions." *Applied Energy* 216 (April): 649–61. https://doi.org/10.1016/j.apenergy.2018.02.110.
- Milligan, Michael, Bethany A. Frew, Aaron Bloom, Erik Ela, Audun Botterud, Aaron Townsend, and Todd Levin. 2016. "Wholesale Electricity Market Design with Increasing Levels of Renewable Generation: Revenue Sufficiency and Long-Term Reliability." *Electricity Journal* 29 (2): 26–38. https://doi.org/10.1016/j.tej.2016.02.005.
- Milligan, Michael, Bethany Frew, Brendan Kirby, Matt Schuerger, Kara Clark, Debbie Lew, Paul Denholm, Bob Zavadil, Mark O'Malley, and Bruce Tsuchida. 2015. "Alternatives No More: Wind and Solar Power Are Mainstays of a Clean, Reliable, Affordable Grid." *IEEE Power and Energy Magazine* 13 (6): 78–87. https://doi.org/10.1109/MPE.2015.2462311.
- Nicolini, Marcella, and Massimo Tavoni. 2017. "Are Renewable Energy Subsidies Effective? Evidence from Europe." *Renewable and Sustainable Energy Reviews*. Elsevier Ltd. https://doi.org/10.1016/j.rser.2016.12.032.
- Parsons Brinckerhoff. 2013. "Electricity Generation Model: 2013 Update of Non-Renewable Technologies." https://www.gov.uk/government/publications/parsons-brinkerhoff-electricity-generation-model-2013-update-of-non-renewable-technologies.
- Pellow, Matthew A., Christopher J.M. Emmott, Charles J. Barnhart, and Sally M. Benson. 2015. "Hydrogen or Batteries for Grid Storage? A Net Energy Analysis." *Energy and Environmental Science* 8 (7): 1938–52. https://doi.org/10.1039/c4ee04041d.
- Pinnangudi, Balasubramanian, Michelle Kuykendal, and Shoham Bhadra. 2017. "Smart Grid Energy Storage." In *The Power Grid: Smart, Secure, Green and Reliable*, 93–135. Elsevier Inc. https://doi.org/10.1016/B978-0-12-805321-8.00004-5.
- Pollitt, Michael G., and Karim L. Anaya. 2016. "Can Current Electricity Markets Cope with High Shares of Renewables? A Comparison of Approaches in Germany, the UK and the State of New York." *Energy Journal* 37 (Special Issue 2): 69–88. https://doi.org/10.5547/01956574.37.SI2.mpol.
- Schmalensee, Richard. 2019. "On the Efficiency of Competitive Energy Storage." *SSRN Electronic Journal*, July. https://doi.org/10.2139/ssrn.3405058.
- Sensfuß, Frank, Mario Ragwitz, and Massimo Genoese. 2008. "The Merit-Order Effect: A Detailed Analysis of the Price Effect of Renewable Electricity Generation on Spot Market Prices in Germany." *Energy Policy* 36 (8): 3086–94. https://doi.org/10.1016/j.enpol.2008.03.035.
- Sepulveda, Nestor A., Jesse D. Jenkins, Fernando J. de Sisternes, and Richard K. Lester. 2018. "The Role of Firm Low-Carbon Electricity Resources in Deep Decarbonization of Power Generation." *Joule* 2 (11): 2403–20. https://doi.org/10.1016/j.joule.2018.08.006.
- Sioshansi, Ramteen, Paul Denholm, Thomas Jenkin, and Jurgen Weiss. 2009. "Estimating the Value of Electricity Storage in PJM: Arbitrage and Some Welfare Effects." *Energy Economics* 31 (2): 269–77. https://doi.org/10.1016/j.eneco.2008.10.005.
- Sisternes, Fernando J. de, Jesse D. Jenkins, and Audun Botterud. 2016. "The Value of Energy Storage in Decarbonizing the Electricity Sector." *Applied Energy* 175 (August): 368–79. https://doi.org/10.1016/j.apenergy.2016.05.014.
- Smith, J.C. 2019. "Meteorology and Market Design For Grid Services Workshop Meeting Overview." Energy Systems Integration Group (ESIG), June 4-6, 2019, Denver, CO. https://www.esig.energy/resources/2019-meteorology-market-design-for-grid-servicesworkshop/.

- Stage, S., and Y. Larsson. 1961. "Incremental Cost of Water Power." Transactions of the American Institute of Electrical Engineers. Part III: Power Apparatus and Systems 80 (3): 361–64. https://doi.org/10.1109/AIEEPAS.1961.4501045.
- Steffen, Bjarne, and Christoph Weber. 2013. "Efficient Storage Capacity in Power Systems with Thermal and Renewable Generation." *Energy Economics* 36 (March): 556–67. https://doi.org/10.1016/j.eneco.2012.11.007.
- Stoft, Steven. 2002. Power System Economics : Designing Markets for Electricity. IEEE Press.
- Strbac, Goran. 2008. "Demand Side Management: Benefits and Challenges." *Energy Policy* 36 (12): 4419–26. https://doi.org/10.1016/j.enpol.2008.09.030.
- Toledo, Olga Moraes, Delly Oliveira Filho, and Antônia Sônia Alves Cardoso Diniz. 2010. "Distributed Photovoltaic Generation and Energy Storage Systems: A Review." *Renewable and Sustainable Energy Reviews*. https://doi.org/10.1016/j.rser.2009.08.007.
- Traber, Thure, and Claudia Kemfert. 2011. "Gone with the Wind? Electricity Market Prices and Incentives to Invest in Thermal Power Plants under Increasing Wind Energy Supply." *Energy Economics* 33 (2): 249–56. https://doi.org/10.1016/j.eneco.2010.07.002.
- Trötscher, Thomas, and Magnus Korpås. 2008. "A Power Market Model for Studying the Impact of Wind Power on Spot Prices." In *16th Power Systems Computation Conference, PSCC 2008.*
- Vita, Alessia De, Izabela Kielichowska, and Pavla Mandatowa. 2018. "Technology Pathways in Decarbonisation Scenarios." *EU-ASSET Project Deliverable*. https://ec.europa.eu/energy/sites/ener/files/documents/2018_06_27_technology_pathways____finalreportmain2.pdf.
- Xu, Bolun, Audun Botterud, and Magnus Korpås. 2019. "Operational Valuation for Energy Storage under Multi-Stage Price Uncertainties," October. http://arxiv.org/abs/1910.09149.
- Xu, Bolun, Alexandre Oudalov, Andreas Ulbig, Göran Andersson, and Daniel S. Kirschen. 2018. "Modeling of Lithium-Ion Battery Degradation for Cell Life Assessment." *IEEE Transactions* on Smart Grid 9 (2): 1131–40. https://doi.org/10.1109/TSG.2016.2578950.
- Zhao, Haoran, Qiuwei Wu, Shuju Hu, Honghua Xu, and Claus Nygaard Rasmussen. 2015. "Review of Energy Storage System for Wind Power Integration Support." *Applied Energy* 137 (January): 545–53. https://doi.org/10.1016/j.apenergy.2014.04.103.

Appendix A: Optimal VRE capacity as a function of investment cost

This appendix provides a detailed analysis of the shape of the VRE capacity curve in Figure 8. Recall from Figure 7 that the net demand curve determines the different price segments. The profitability of VRE depends on how much production that takes place in the different price segments. Lower investment cost leads to higher VRE capacity which again alters the net demand curve.

In Figure 15 we have plotted the duration curve for net demand together with the corresponding VRE output for different VRE investment cost scenarios. With an investment cost of 4800 €/kW (blue curve), VRE is only marginally profitable, and the installed capacity is so low that the net demand is almost identical to the demand curve without VRE. We see that the VRE output variations sorted by the net demand follows a rather favorable pattern, with slightly higher production during high demand. This result is due to the weather variations of European offshore wind with highest average wind speeds during winter when the demand also is high. With a slight decrease in investment cost (red curve), the optimal installed VRE capacity increases, and pushes the net demand curve downwards. Consequently, more VRE production takes place during lower net demand. The VRE capacity is still so low that there are no hours with zero prices. When the investment cost is reduced to 4500 €/kW (yellow curve), the installed VRE capacity has reached a level where situations with price equal to zero are about to occur. A further decrease in investment cost from this point is expected to lead to marginally less added VRE capacity due to two reasons: 1) Higher VRE capacity leads to more and more hours with zero prices, 2) Higher VRE capacity alters the net demand slightly and thereby pushes even more of the VRE production to the low-price hours.

When the investment cost is reduced below 2000 €/kW, the installed VRE capacity is already at such a high level that a continued increase in capacity does not noticeably change the VRE output variations in the different price segments (the green and purple VRE output curves are almost identical). This "saturation" of VRE variations gives rise to an interesting effect on the installed VRE capacity, which from this point on increases non-linearly again, as shown in Figure 7. This result can be explained from the cost recovery condition of the VRE plant from Chapter 4.1:

$$\pi_{\nu} = 0 \Rightarrow F_{\nu} = \nu_s \int_0^{t_s} AF_{\nu}(t)dt + \nu_p \int_{t_s}^{t_p} AF_{\nu}(t)dt + \nu_b \int_{t_p}^{t_b} AF_{\nu}(t)dt$$

where AF_{v} refers to the Availability Factor of the VRE plant before curtailment. From Figure 15, we see that AF_{v} can be approximated by a step-wise linear function of t for VRE investment cost of 1000 \notin /kW (green) and 2000 \notin /kW (purple):

$$F_{v} \approx v_{s} \int_{0}^{t_{s}} (a_{1}t + b_{1})dt + v_{p} \int_{t_{s}}^{t_{p}} (a_{2}t + b_{2})dt + v_{b} \int_{t_{p}}^{t_{b}} (a_{3}t + b_{3})dt$$

$$= \frac{1}{2} v_{s} a_{1} t_{s}^{2} + \frac{1}{2} v_{p} a_{2} (t_{p} - t_{s})^{2} + \frac{1}{2} v_{b} a_{3} (t_{b} - t_{p})^{2} + v_{s} t_{s} b_{1} + v_{p} b_{2} (t_{p} - t_{s})$$

$$+ v_{b} b_{3} (t_{b} - t_{p}) \Rightarrow F_{v} = c_{1} t_{b}^{2} - c_{2} t_{b} + c_{3}$$

where a_i, b_i, c_i refers to constants. As explained in the main part of the paper, the optimal values of t_s and t_p remains constant as long as the peaker and baseplant are not entirely pushed out of the system. The cost recovery condition gives thus a quadratic relation between cost-optimal duration of the baseload plant t_b and VRE investment cost if the availability factor of VRE, AF_v , is a linear function of the net demand duration t. The relation between t_b and VRE capacity, x_v , is given by the net demand curve. With an approximated linear relation between net demand duration and VRE output, we can also express the net demand as a linear function of VRE capacity x_v : $q_{nd}(t) \approx a_6 - a_7 \cdot t \cdot x_v$. The base duration t_b is given by $q_{nd}(t_b) = 0 \Rightarrow t_b = a_8/x_v$. Inserting into the cost-optimality condition above, we obtain the following non-linear relation between VRE investment cost and VRE capacity:

$$F_v = Ax_v^{-2} - Bx_v^{-1} + C$$

where A, B and C are constants. This function approximates the right part of Figure 8, with the axes inverted. For very high x_v , the investment cost goes asymptotically towards the constant C, which we also can observe from Figure 8. As F_v goes towards zero, the VRE capacity will go towards the amount that is required to supply the load by VRE only. This is theoretically possible if the VRE output is always higher than zero, which e.g. can be the case for aggregation of VRE over a large region or country.



Figure 15. Duration curves of net demand (lower) and corresponding VRE output sorted by net demand (daily averages, upper) for different VRE investment cost scenarios. The corresponding VRE installed capacities and relative VRE curtailments are 3.6 MW / 0 % (blue), 19.4 MW / 0 %(red), 48.2 MW / 0.001 % (yellow), 93.0 MW / 8.9 % (purple), 121.1 MW / 22 % (green).



MIT Center for Energy and Environmental Policy Research

Since 1977, the Center for Energy and Environmental Policy Research (CEEPR) has been a focal point for research on energy and environmental policy at MIT. CEEPR promotes rigorous, objective research for improved decision making in government and the private sector, and secures the relevance of its work through close cooperation with industry partners from around the globe. Drawing on the unparalleled resources available at MIT, affiliated faculty and research staff as well as international research associates contribute to the empirical study of a wide range of policy issues related to energy supply, energy demand, and the environment.

An important dissemination channel for these research efforts is the MIT CEEPR Working Paper series. CEEPR releases Working Papers written by researchers from MIT and other academic institutions in order to enable timely consideration and reaction to energy and environmental policy research, but does not conduct a selection process or peer review prior to posting. CEEPR's posting of a Working Paper, therefore, does not constitute an endorsement of the accuracy or merit of the Working Paper. If you have questions about a particular Working Paper, please contact the authors or their home institutions.

MIT Center for Energy and Environmental Policy Research 77 Massachusetts Avenue, E19-411 Cambridge, MA 02139 USA

Website: ceepr.mit.edu

MIT CEEPR Working Paper Series is published by the MIT Center for Energy and Environmental Policy Research from submissions by affiliated researchers.

Copyright © 2020 Massachusetts Institute of Technology For inquiries and/or for permission to reproduce material in this working paper, please contact:

 Email
 ceepr@mit.edu

 Phone
 (617) 253-3551

 Fax
 (617) 253-9845